# rivista di statistica ufficiale

## **REVIEW OF OFFICIAL STATISTICS**

In this issue:



A systematic and standardised burden measurement system for surveys on businesses Annamaria D'urzo, Angela Golino, Stefania Macchia, Alessandra Nurra, Carmela Pascucci, Mariagrazia Rinaldi, Fabiana Rocci, Carla Schiattone, Giorgia Simeoni

Handling Multinational Enterprises: Istat's early experience in Large Cases Unit Simone Ambroselli, Sonia Amante, Elisabetta Bilotta, Barbara Gentili, Barbara Iaconelli, Luigia Mattei, Chiara Orsini, Fabiana Sartor, Ilaria Straccamore

The task definition process for Istat Large Cases Unit Simone Ambroselli, Sonia Amante, Elisabetta Bilotta, Barbara Gentili, Chiara Orsini, Fabiana Sartor



# rivista di statistica ufficiale

## **REVIEW OF OFFICIAL STATISTICS**

#### In this issue:

### n. 1 2019

A systematic and standardised burden measurement system	
for surveys on businesses	
Annamaria D'urzo, Angela Golino, Stefania Macchia,	
Alessandra Nurra, Carmela Pascucci, Mariagrazia Rinaldi,	
Fabiana Rocci, Carla Schiattone, Giorgia Simeoni	9
Handling Multinational Enterprises: Istat's early experience in Large Cases Unit	
Simone Ambroselli, Sonia Amante, Elisabetta Bilotta,	
Barbara Gentili, Barbara Iaconelli, Luigia Mattei,	
Chiara Orsini, Fabiana Sartor, Ilaria Straccamore	51
The task definition process for Istat Large Cases Unit	

Ine task definition process for Istat Large Cases UnitSimone Ambroselli, Sonia Amante, Elisabetta Bilotta,Barbara Gentili, Chiara Orsini, Fabiana Sartor85



#### Editor: Patrizia Cacioli

#### Scientific committee

**President:** Gian Carlo Blangiardo

#### Members:

vicinities.			
Corrado Bonifazi	Vittoria Buratta	Ray Chambers	Francesco Maria Chelli
Daniela Cocchi	Giovanni Corrao	Sandro Cruciani	Luca De Benedictis
Gustavo De Santis	Luigi Fabbris	Piero Demetrio Falorsi	Patrizia Farina
Jean-Paul Fitoussi	Maurizio Franzini	Saverio Gazzelloni	Giorgia Giovannetti
Maurizio Lenzerini	Vincenzo Lo Moro	Stefano Menghinello	Roberto Monducci
Gian Paolo Oneto	Roberta Pace	Alessandra Petrucci	Monica Pratesi
Michele Raitano	Maria Giovanna Ranalli	Aldo Rosano	Laura Terzera
Li-Chun Zhang			

#### **Editorial board**

**Coordinator:** Nadia Mignolli

#### Members:

- Ciro Baldi Tania Cappadozzi Roberto Colotti Alessandro Faramondi Luisa Franconi Filippo Moauro Federica Pintaldi Isabella Siciliani Francesca Vannucchi
- Patrizia Balzano Anna Maria Cecchini Stefano Costa Francesca Ferrante Antonella Guarneri Filippo Oropallo Maria Rosaria Prisco Marina Signore Claudio Vicarelli
- FedericoBenassi Annalisa Cicerchia Valeria De Martino Maria Teresa Fiocca Anita Guelfi Alessandro Pallara Francesca Scambia Francesca Tiero Anna Villa

Giancarlo Bruno Patrizia Collesi Roberta De Santis Romina Fraboni Fabio Lipizzi Laura Peci Mauro Scanu Angelica Tudini

#### **rivista di statistica ufficiale** n. 1/2019

Four-monthly Journal: registered at the Court of Rome, Italy (N. 339/2007 of 19th July 2007).

e-ISSN 1972-4829 p-ISSN 1828-1982 © 2022 Istituto nazionale di statistica Via Cesare Balbo, 16 – Roma



Unless otherwise stated, content on this website is licensed under a Creative Commons License - Attribution - 3.0. <u>https://creativecommons.org/licenses/by/3.0/it/</u> Data and analysis from the Italian National Institute of Statistics can be copied, distributed, transmitted and freely adapted, even for commercial purposes,

provided that the source is acknowledged.

No permission is necessary to hyperlink to pages on this website. Images, logos (including Istat logo), trademarks and other content owned by third parties belong to their respective owners and cannot be reproduced without their consent.

The views and opinions expressed are those of the authors and do not necessarily reflect the official policy or position of the Italian National Institute of Statistics - Istat.

The Scientific Committee, the Editorial Board and the authors would like to thank the anonymous reviewers (at least two for each article, on a voluntary basis and free of charge, with a double-anonymised approach) for their comments and suggestions, which enhanced the quality of this issue of the Rivista di statistica ufficiale.

#### **Editorial Preface**

The three scientific articles published in this issue of the *Rivista di statistica ufficiale* deal with business statistics.

In the first study, Annamaria D'urzo, Angela Golino, Stefania Macchia, Alessandra Nurra, Carmela Pascucci, Mariagrazia Rinaldi, Fabiana Rocci, Carla Schiattone and Giorgia Simeoni illustrate the commitment carried out by an internal Working Group of the Italian National Institute of Statistics – Istat.

The main purpose is to identify, process and implement a representative set of indicators and the reference methodology for measuring the respondent burden to be applied to business surveys.

An additional objective is to release them in a systematic and standardised way through a generic procedure.

The two following articles are complementary to each other and concern Large Cases Units.

The first one is written by Sonia Amante, Simone Ambroselli, Elisabetta Bilotta, Barbara Gentili, Barbara Iaconelli, Luigia Mattei, Chiara Orsini, Fabiana Sartor and Ilaria Straccamore, with the purpose of examining the way Multinational Enterprises organise their businesses worldwide and the consequent impact on the production of official statistics.

In 2016, the modernisation process started at Istat, also enhancing an *ad hoc* team to support Large Cases Units. The main results of the several analyses conducted during the first phase of this project are also documented in the present paper, evaluating size, significance and complexity aspects related to the characteristics of the national economy.

Size aspect defined according to several variables, *e.g.* gross value added, turnover, value of production or number of employees, resulted very useful for the identification of the largest Multinational Enterprises. In the meantime, complexity needs to be evaluated on the basis of other features such as the enterprise group structure and the number of activities carried out.

Finally, significance for data collection is related to the selection of the most important respondents, *i.e.* the dominant groups or those who have data provision problems.

In the closing paper of this issue, Sonia Amante, Simone Ambroselli, Elisabetta Bilotta, Barbara Gentili, Chiara Orsini and Fabiana Sartor continue to highlight the relevance of Large Cases Units both within National Institutes and for official statistics, focussing on their multidimensional meaning.

More specifically, they describe the experience of the Italian National Institute of Statistics – Istat together with the steps followed in defining the operational tasks needed to detect complex procedures involving Multinational Enterprises.

All these phases have an impact both on data quality and on international comparability.

Nadia Mignolli Coordinator of the Editorial board

# A systematic and standardised burden measurement system for surveys on businesses

Annamaria D'urzo, Angela Golino, Stefania Macchia, Alessandra Nurra, Carmela Pascucci, Mariagrazia Rinaldi, Fabiana Rocci, Carla Schiattone, Giorgia Simeoni<sup>1</sup>

#### Abstract

Following what emerged from the second round of the peer reviews on the ES Code of Practice carried out in 2015, Istat decided to propose a systematic and standardised burden measurement system, to be implemented for surveys on business through the web-based data collection.

This paper describes the study conducted to identify a representative set of indicators and the methodology to elaborate them. The results of a test run on several businesses surveys are also described and some reflections about possible strategic actions to contain burden are presented.

Keywords: Response burden, business surveys.

<sup>1</sup> Annamaria D'Urzo (<u>adurzo@istat.it</u>); Angela Golino (<u>golino@istat.it</u>); Stefania Macchia (<u>macchia@istat.it</u>); AlessandraNurra(<u>nurra@istat.it</u>); CarmelaPascucci(<u>pascucci@istat.it</u>); MariagraziaRinaldi(<u>mrmarina@istat.it</u>); Fabiana Rocci (<u>rocci@istat.it</u>); Carla Schiattone (<u>schiatto@istat.it</u>); Giorgia Simeoni (<u>simeoni@istat.it</u>), Italian National Institute of Statistics - Istat.

The views and opinions expressed are those of the authors and do not necessarily reflect the official policy or position of the Italian National Institute of Statistics - Istat.

The authors would like to thank the anonymous reviewers for their comments and suggestions, which enhanced the quality of this article.

#### 1. Introduction<sup>2</sup>

This paper describes a study carried out by an internal Working Group<sup>3</sup> of the Italian National Institute of Statistics – Istat aimed at defining and implementing a system of indicators for the measurement of respondent burden for surveys on businesses.

One of the recommendations resulting from the second round of peer reviews on the implementation of the ES Code of Practice<sup>4</sup>, underlined that Istat was already collecting information on response burden in some business surveys but not in a systematic and standardised way. Thus an improvement action for the definition and implementation of a systematic and standardised burden measurement system was formulated by Istat. The Working Group, set up at the end of 2015, had the aim of complying with such improvement action.

The purpose of this report is to describe the set of indicators on response burden that Istat has designed for surveys on businesses in a strategic perspective, so as to release them in a systematic and standardised way through a generalised software procedure. The results obtained on a first set of three surveys are presented, and some reflections about the possible strategic actions to contain burden are mentioned.

The following Section 2 specifies some context information related to the modernisation of the production process that Istat has been udergoing since 2016 (Istat, 2016). Section 3 reports some general concepts and definitions on response burden measurements and the general decisions endorsed by the Working Group on what kind of burden will be measured and how. Section 4 presents the set of indicators that the Working Group has defined in order to measure the response burden by survey and Section 5 shows the preliminary results obtained on three surveys, on which the standardised process was tested according to their different features (short term / structural survey, short/long

<sup>2</sup> Although the article is the result of a joint work, the single parts are authored as follows: Sections 1, 2, 3, and 4 by Stefania Macchia; Paragraph 5.1 by A. Nurra; Paragraph 5.2 by A. Golino and F. Rocci; Paragraph 5.3 by C. Pascucci and F. Rocci; Section 6 by M. Rinaldi and C. Schiattone; Section 7 by A. D'Urzo; Section 8 by G. Simeoni.

<sup>3</sup> Coordinators: S. Macchia, N. R. Fazio. Members: S. Cuomo, A. Golino, R. Mazzucco, M. Murgia, A. Nuccitelli, A. Nurra, A. Nunnari, P. Papa, C. Pascucci, S. Pietropaoli, M. Rinaldi, F. Rocci, C. Schiattone, G. Simeoni.

<sup>4</sup> For further information on the peer reviews on the ES Code of Practice please refer to <a href="http://ec.europa.eu/eurostat/web/quality/peer-reviews">http://ec.europa.eu/eurostat/web/quality/peer-reviews</a>.

questionnaires). Section 6 describes the Automatic Procedure for Burden Indicators by Survey which implements the standardised process, while Section 7 is devoted to the percistency indicators. Finally, some preliminary suggestions to contain burden are identified.

#### 2. The transition to the new model for business surveys

It is worth mentioning that since 2016 Istat, in the context of the newly established Business Architecture Activity Model (Istat, 2016), has been reunifying all the scattered resources and processes pertaining to the data collection phase under a new dedicated Directorate. Thus extending the principles underpinning the reorganisation which the business survey system has undergone in recent years. The main goal of that reorganisation was to abandon the so-called 'stovepipe' model, in which each 'stovepipe' identifies a specific field of statistics and its corresponding production system, to implement a new model in which the production of statistics is no longer expected to operate through independent processes, but rather as a single, consistent and integrated pool of non-redundant building blocks (enterprise-centred model).

Cornerstone of this new system is the Istat Business Statistical Portal (N. Fazio, M. Murgia, A. Nunnari, 2013) an integrated system for the management of data collection processes, which is, at the same time, an attempt to streamline the organisation and production processes of business surveys as a whole.

The Portal acts as a single entry point for web-based data collection from enterprises, according to a 'business-centric' perspective. It provides new integrated functions supporting respondents in several areas: survey unit management and updating, data collection process management, direct and proxy compilation of electronic questionnaires. The environment also provides a software tool to develop/design electronic questionnaires: GX (Generalised Italian (Data) Collection System XML), an in-house product using XML, to represent the main survey's structure, *i.e.* survey metadata, survey variables, questionnaire structure, check plan and skipping rules.

To fully achieve its goals, this new architecture relies heavily on fasttracking the semantic and syntactic harmonisation of survey questionnaires, *i.e.* both in terms of concepts and design, with the perspective of improving quality and containing the respondent burden.

Up to now, surveys questionnaires were implemented with different, often dedicated/*ad hoc* software systems, so the migration of all of them in the Business Portal data collection system is gradually undergoing.

This emerging context of integration and common IT solutions is encouraging the definition of standards and recommended practices for questionnaire design and for all functions belonging to the data collection phase, including the sets of indicators aimed at monitoring and enhancing quality of the data collection processes.

#### 3. The burden to be measured

The literature on the measurement of statistical burden on businesses is quite wide and a review of it can be found in the Memobust handbook (2014, Eurostat). Building on this, in order to identify indicators to be produced, it was needed to provide some details to better substantiate what was intended to measure. It is possible to narrow down the factors that contribute to create burden to two main classes:

- actual/objective factors, mainly due to time spent to provide responses;
- subjective factors, connected with what is 'perceived' as burden by respondents.

The first kind of factors can be measured in terms of time, but also of costs. For this purpose, a model has been evaluated, the so called SCM - Standard Cost Model (2004, Eurostat). On balance, this was not considered as a viable, cost-effective option, as SCM requires analysts to collect a broader set of information than just time spent completing the questionnaire. (2012J.Jones).

On the other hand, the concept of time spent to provide responses needs to be specified as well, making clear what it is intended to include. In particular, it should be decided whether considering all the following activities:

- time to understand what is being required;
- time to retrieve data to be provided;
- time to complete the questionnaire;
- time to respond to re-contacts, *e.g.* during the data editing phase.

Istat decided that the burden each survey places on respondent businesses will be measured only in terms of: *i*) time to retrieve information to be provided; *ii*) time to completely fill in the questionnaire. It was considered that tracking time spent on the remaining activities in the list would have been too complex and time consuming.

The identification of the 'perceived burden' is even more complex as it could depend on a number of aspects, such as the survey design, the respondent's characteristics and other external factors, that might require gathering additional data directly from the respondents. In designing the set of indicators to measure burden, Istat sought a solution imposing two constraints:

- do not cause further burden to respondents in order to collect information on burden;
- limit internal investments to set up IT procedure to estimate burden in different surveys.

The first constraint compelled to define a set of indicators by relying on a strategy of maximum exploitation of available sources and minimal request of information from respondents.

Following the second constraint, it has been decided that **data concerning burden will be collected and processed for surveys already residing in the Portal environment or as soon as they migrate to it, so as to implement and set up a generalised software procedure**.

Besides what already mentioned, the burden can be considered from two different perspectives:

- burden by survey (BBS), *i.e.* the burden the single survey places on the involved businesses;
- burden by business (BBB), *i.e.* the total amount of burden generated by all the surveys a business is involved in.

#### Istat decided to measure burden from both points of view, expressing BBS in terms of total time spent to fulfill the requested survey task, and BBB in terms of persistency.

The choice not to produce burden indicators for business surveys that still are not migrated to the new environment is offset by means of the persistency indicators, which will allow to have an overall view of burden imposed on all businesses surveyed by Istat.

The last aspect which has to be stressed before describing the proposed indicators is that they are aimed at representing the evolution of the phenomenon of burden more than a precise estimate of a statistical entity. This is because the indicators are thought for all kinds of surveys (both short term and structural ones), that involve different types of sample designs and different treatments of changes that occur to the respondent units during the same year. Hence, the chosen criteria for selecting the set of businesses to be considered in each period to measure burden will not take into account a series of events (for instance events depending on the business demography, as well as late responses of businesses providing data after the given deadline) which, on the contrary, are relevant for the survey's results. This is to assure the comparability of the several surveys burden indicators to be evaluated year by year.

#### 4. Burden indicators by survey (BBS)

As already mentioned, the burden that each survey lays on respondent businesses will be measured in terms of time to complete the questionnaire and time to retrieve information needed. In particular, burden will be considered for all the respondent businesses (businesses who submitted the filled-in questionnaire). In details:

- **time to fill in the questionnaire** will be quantified processing the paradata<sup>5</sup> the data collection system automatically records, indeed it was decided to exploit sources of information automatically generated by the software system (paradata) and minimise the request of data to businesses. As known, paradata are a rich informative source for monitoring the data collection process (M.P. Couper, 1998), as they store automatically all actions performed by respondents while navigating the data collection environment;
- time to retrieve information to provide data and other information will be asked to respondents in an *ad hoc* section to be added to each surveys questionnaire as they migrate to the Business Portal system (the '*Burden section*' is shown in Figure 4.1).

There is a very important difference between the two sources used: paradata are exhaustive, which means they are available on all businesses who submitted the filled-in questionnaire, while data coming from the *Burden section* of the questionnaires are partial as answering was not compulsory. On the other hand, it has been confirmed (see Section 5) that the set of businesses which fills in this section is not particularly characterised against the complete set of respondent businesses.

<sup>5</sup> The paradata of a data set or survey are data about the process by which the data were collected. Example paradata topics about a survey include the times of day interviews were conducted, how long filling in the questionnaire took, whether questionnaire was completed with a single access to the system or in different times, etc. Thus there are paradata about each 'observation' in the survey. The analysis of these data are useful to asses the costs and management of a survey, so as to identify possible improvements of the questionnaire design.

#### Figure 4.1 - The Burden section

2. Indica questi Please cl	tte the time you onnaire, selection neck only one ansy	spent to retrieve ng one of the fo ver	e the informatio llowing time clo	on necessary to asses	fill in the		
1	2	3	4	5	6	7	The upper classes won
o to 15 minutes	From 15 minutes to half an hour	More than half an hour and up to 1 hour	More than 1 hour and up to 3 hours	More than 3 hours and up to 10 hours	More than 10 hours and up to 30 hours	More than 30 hours	be presented for very shor questionnair
							-
3. Indica the qu More ta	tte whether and uestionnaire: han an answer is p	which difficulti	ies you encount	ered in filling i	n		
3. Indica the q More ta	ate whether and uestionnaire: han an answer is p a) Technical o	which difficulti nossible 1nd IT problems	ies you encount	ered in filling i	<b>n</b>		
3. Indica the q More t	tte whether and uestionnaire: han an answer is p a) Technical a b) Difficulties	which difficulti nossible and IT problems in navigating th	ies you encount	ered in filling is	n 	Not pertinent f	ior
3. Indica the q. More t	tte whether and uestionnaire: han an answer is p a) Technical o b) Difficulties c) Too many o	which difficulta nossible and IT problems in navigating th questions	ies you encount	ered in filling i	n 	Not pertinent f very short questionnaires	for
3. Indica the q. More t	<ul> <li>a) Technical of</li> <li>b) Difficulties</li> <li>c) Too many of</li> <li>d) Insufficient</li> </ul>	which difficulti nossible and IT problems in navigating th questions ty clear question	ies you encount	ered in filling i	n ns	Not pertinent f very short questionnaires	for
3. Indica the q. More t	<ul> <li>a) Technical a</li> <li>b) Difficulties</li> <li>c) Too many a</li> <li>d) Insufficient</li> <li>e) Unclear or</li> </ul>	which difficulti nossible and IT problems in navigating th questions ly clear question not exhaustive of	ies you encount hrough the ques ns	ered in filling i	ns	Not pertinent f very short questionnaires	for
3. Indica the q. More t	<ul> <li>the whether and uestionnaire:</li> <li>than an answer is p</li> <li>a) Technical a</li> <li>b) Difficulties</li> <li>c) Too many a</li> <li>d) Insufficient</li> <li>e) Unclear or</li> <li>f) Not enough questions</li> </ul>	which difficult nossible and IT problems in navigating th questions ty clear question not exhaustive of a information su	ies you encount hrough the ques ns answer options pporting the un	ered in filling is tionnaire sectio derstanding of	ns	Not pertinent f very short questionnaires	ior
3. Indica the q. More to	<ul> <li>the whether and uestionnaire:</li> <li>than an answer is p</li> <li>a) Technical d</li> <li>b) Difficulties</li> <li>c) Too many d</li> <li>d) Insufficient</li> <li>e) Unclear or</li> <li>f) Not enough questions</li> <li>g) Excessive a information</li> </ul>	which difficulta nossible and IT problems in navigating the questions ty clear question not exhaustive of n information sup listance between n available in th	ies you encount hrough the ques ns answer options pporting the un information re e business	ered in filling is tionnaire sectio derstanding of quested and	n	Not pertinent f very short questionnaires	for

Source: Istat - Study on the burden measurement system

#### 4.1 Time spent to fill in the questionnaire

As already mentioned, paradata will be used to estimate the time spent to fill in the questionnaire. The strength of using this source is that it is absolutely objective and precise. It allows to quantify the actual and net time each respondent dwells on each of the questionnaire sections, even if the compilation takes place in different sessions. As a matter of fact, with paradata the start and end times of each compilation session are recorded (date, hour, minute, second). By processing these timestamps, it is possible to measure the duration of each session and, in case the questionnaire has been filled during different sessions, the summed duration of all sessions represents the total actual time spent to fill in the questionnaire. As we see, this calculation is based on primary data and absolutely objective, while compilation times elicited directly from respondents could be affected by subjective evaluation and prone to perception bias.

By processing paradata it is also possible to get additional and very interesting information, for instance: the gross amount of time the respondent spent connected to the web questionnaire (from the start date and time of the first session to the end date and time of the last session), whether the respondent did it in one single session or in several sessions, whether he/she made it in a single day or across several days.

The indicator of burden will be processed for respondent businesses, namely those that submitted a completed questionnaire. This does not imply any further consideration for structural surveys (SBS), while a clarification has to be made on short term surveys (STS): only businesses who provided data for all the periods of the reference year are considered as respondents (12 months for monthly surveys, 4 quarters for quarterly surveys).

So the indicators, for SBS and STS surveys respectively, are as follows:

#### SBS surveys

(1) **ACT** (Average Compilation Time) =  $\sum_{i=1}^{n} (CT_i)/n$ 

where

 $CT_i$  = compilation time, expressed in minutes, of the 'i' respondent business

n = total number of respondent businesses

**STS surveys** 

(2) 
$$\overline{ACT} = \frac{\sum_{j=1}^{m} (\sum_{i=1}^{n} (CT_{ij})/n)}{m}$$

where

 $CT_{ij}$  = compilation time, expressed in minutes, of the 'i' respondent business in the 'j' period

n = total number of businesses respondent for all the periods of the reference year (constant value for all the periods)

m = total number of periods (periods of the reference year for which the respondents provided data, which means 12 months for monthly surveys, 4 quarters for quarterly surveys)

As it can be seen from the formulas, this indicator is a simple mean, but it provides survey managers with a starting point for further analysis, for instance to highlight whether there are significant differences depending on the businesses' characteristics (in terms of dimension or turnover or economic activity sector).

# 4.2 Time spent retrieving required information and number of people involved in the task

The answer given by respondents to the second question of the *Burden section* will be used to estimate time spent to retrieve information. This datum is surely subjective and, if the respondent perceives the task of providing data for statistical survey as a burden, it could be overestimated. On the other hand, the experience in conducting business survey clearly shows that within the 'Response process' (Edwards and Cantor, 1991; Sudman *et al.*, 2000; Willimack and Nichols, 2001) the step of 'retrieval information' is particularly heavy for businesses because it often implies finding and analysing data from corporate databases and/or paper filing systems. It follows that this information must absolutely be taken into account in estimating the respondent burden.

As it can be seen in Figure 4.1, the question in the *Burden section (time you spent to retrieve information necessary to fill in the questionnaire)* is structured in classes (class 1: up to 15 minutes; class 2: from 15 minutes to half an hour, etc.). An approximation of the time spent on average will be

elaborated, taking the central value of each class as reference time for the final calculus. The last class will not be considered for this calculus because defining the central value would need not available information. In addition, this class is selected by a very low percentage of respondents (see Section 5).

In synthesis, it will be expressed through the following indicator:

(3) **ART** (Average Retrieval Time) =  $\sum_{i=1}^{n} (RT_i)/n$ 

where

 $RT_i$  = retrieval time of the 'i' respondent business

n = total number of businesses which provided this information in the *Burden section* 

Another aspect was considered relevant to estimate burden: the number of people involved in providing information. It is worth considering that the fact of involving more than one person may have several explanations: providing the requested data requires different professional skills, so that different experts within the business must be contacted or, particularly for STS surveys, the person who fills the questionnaire could not be the same in all data collection periods during the year. Anyway, collecting such an information can be helpful for the survey manager to conduct further analysis.

The indicator will be expressed through the following formula:

(4)  $\overline{PI}$  (Persons Involved) =  $\sum_{i=1}^{n} (PI_i)/n$ 

Where

 $PI_i$  = number of persons involved in providing information of the 'i' respondent business

n = total number of businesses which answered to this question in the *Burden section* 

As already said, the *Burden section* was added to questionnaires as surveys migrated to the new GX system. It must be specified that for STS surveys, it was decided to request the *Burden section* only in the last period (last month or last quarter) of the collection year, specifying that the information requested was the average time spent to retrieve information to fulfill the task in the different periods of the year.

# 4.3 Possible burden caused by difficulties encountered in providing information

With the third question of the *Burden section (Indicate whether and which difficulties you encountered in filling in the questionnaire*) some aspects related to other possible causes of burden pertaining to survey design are investigated: questionnaire length, questionnaire design (in terms of clarity of questions, support information, question options), usability of the electronic questionnaire (ease of navigation and functions supported), IT problems, etc. The set of response options of this question is surely not exhaustive and respondent's attitude towards collaboration is not being investigated, but this would have required a wider set of questions which in turn would have caused more burden on respondents. Certainly this question does not allow to single out the specific problem/difficulty (if the option '*unclear and not exhaustive questions*' is selected, it is not known which question it is referred to), but it can be viewed as evidence of some problems occurring, especially in the context of surveys recently migrated to the new IT system, or every time the questionnaire has undergone deep changes.

The indicator will be simply expressed through a table showing the frequency distribution of businesses per number of problems encountered. This information allows to monitor the evolution of the phenomenon, with the aim of reducing the percentage of respondents encountering higher numbers of difficulties.

Number of difficulties declared by the respondent	Number of respondent businesses	% of respondent businesses
0		
1		
2		
3		

#### Table 4.1 – Respondents

Source: Istat - Study on the burden measurement system

#### 4.4 Summary report on burden by survey (BBS)

For each survey the defined set of indicators will be systematically calculated.

Besides the values of the indicators, some further information on the survey, on the questionnaire and on the actions undertaken to reduce the respondents' burden can be of interest for an overall assessment of the response burden by survey.

First of all, the raw<sup>6</sup> number of respondents (respondent businesses) to the survey provides an idea of the survey size and the 'incidence' of the burden on the businesses population.

Secondly the raw number of respondent units is reported because the burden is measured at business level, but a business, if articulated in different units, would provide information for all of them, so knowing the number of units pertaining to a single business helps to explain the obtained value of the burden indicator.

Finally it can be interesting to know what actions have been already implemented to contain the burden, in particular if the sample was selected limiting the overlapping with other surveys.

The following figure summarises the information to be provided/collected for each survey.

<sup>6 &#</sup>x27;Raw' meaning that businesses responding after the deadline, for example, are not considered.

#### Figure 4.2 - Summary report



Source: Istat - Study on the burden measurement system

In addition, another information can be analysed to explain burden, even if it will not be included in the summary report because mostly relevant for internal experts: the level of complexity of the questionnaire. Istat has defined a simple indicator to measure in a standard and comparable way the complexity of its questionnaires. The indicator of complexity (CI)<sup>7</sup> takes into account the number and type of questions included in the questionnaire, the difference between the minimum and the maximum number of answers that should be provided and the presence of routing rules. These different factors are then summarised on a qualitative scale (easy, medium, difficult). The value of the CI can be used as a support to compare the indicators on burden across different surveys: e.g. if two surveys have the same level of CI and very different values for the burden indicators the situation calls for further investigation. The difference in topics between the two surveys can, for example, justify the variation, but it is also possible that the survey with higher level of burden would benefit from some improvements in its design to reduce it.

<sup>7</sup> The concepts considered for this index are: number of questions, scores of questions and paths to fill in the questionnaire. Different scores are assigned according to the characteristics of the questions (open-ended, closed-ended, multiple choice, etc.). The different paths of the questionnaires are analysed in order to identify the shortest and the longest path.

Questions scores corresponding to the shortest and to the longest path are calculated.

The Index is the arithmetic average between these two scores.

#### 5. Results of burden indicators in ICT, LES and Turnover surveys

It is necessary to mention that those reported in this document are the first results obtained through the described methodology, which could be further assessed after a deeper analysis. In particular more insights and reflections are foreseen for short terms surveys to validate the methodologic choices defined for them through the analysis of the first year results.

#### 5.1 First results of burden indicators on ICT survey

Istat conducts survey on the use of information and communication technologies (ICT) on the basis of EC Regulation on Statistics on Information society involving active enterprises in industry and services with 10 or more employees. Since the year 2014 this survey has been managed through the new data collection system integrated into the Business Statistical Portal. Concurrently, major/significant innovations were introduced, regarding the overall design of the questionnaire, the way inconsistencies are highlighted in the electronic form when rules violations are triggered and the tools for monitoring the survey progress. Moreover, it was decided to add a new section at the end of the questionnaire to measure *respondent burden* in terms of classes of time necessary to find information (*retrieval time*) and about number of persons involved in providing the requested information.

In the rest of the present paragraph, results of the analysis on both the *Burden section* and the compilation time are presented.

Response rate of ICT survey in 2015 was about 61%. Table 5.1.1 reports the percentage distribution of respondents who only answered the survey and respondents who gave an answer also to the *Burden section*, showing a very good data representativeness.

Size class	Total	Respondent to ICT and not to burden	Respondent both to ICT and burden	
		percenta	ercentages	
10-19	9,146	0.70	99.30	
20-49	3,675	0.65	99.35	
50-99	1,982	0.50	99.50	
100-249	2,109	0.95	99.05	
250+	2,509	0.88	99.12	
Total	19,421	0.72	99.28	

Table 5.1.1 - Respondents ICT2015: Bu	rden section by size class (absolute value and
percentages)(a)	

Source: Istat – ICT survey: study on the burden measurement system

(a) Enterprises are considered as respondents to the *Burden section* if they answered at least one question between retrieval time and persons involved.

Table 5.1.2 shows the retrieval of information time classes by size classes of enterprises. As we see, there is a clear-cut concentration of enterprises of all sizes in the first three classes of retrieval time and, moreover, it is evident that a larger amount of time was needed by companies with at least 50 persons employed (*i.e.* more than 60% of large enterprises needs a time between 30 minutes and 3 hours to find information). The extra time required by larger enterprises is justified by the fact that a more complex organisation implies more intensive use of ICT, which results in a longer path of the questionnaire (different paths are due to the responses to the filter questions).

Table 5.1.2 - Respondents ICT2015: time to retrieve information by size class (percentages)(a)

	Time to retrieve information					
Size class	Up to 30 mi- nutes	More than half an hour and up to 1 hour	More than 1 hour and up to 3 hours	More than 3 hours and up to 10 hours	More than 10 hours and up to 30 hours	More than 30 hours
10-19	41.11	41.00	15.77	1.58	0.29	0.25
20-49	36.64	41.95	18.36	2.53	0.28	0.25
50-99	32.49	39.86	22.01	4.73	0.51	0.41
100-249	25.11	38.36	28.52	6.05	0.91	1.06
250+	15.41	35.94	34.33	10.85	1.82	1.65
Total	34.33	40.12	20.68	3.76	0.57	0.54

Source: Istat – ICT survey: study on the burden measurement system (a) Missing: 210.

Similarly, with increasing firm size also the number of people involved in compiling increases due to the greater variety of skills required to answer questions of different topics investigated by ICT survey (Table 5.1.3). Compared with an average of about 1.7 people needed by enterprises with fewer than 20 persons employed, filling in the model takes on average 3.5 persons to the larger enterprises with almost 250 persons employed. The more general *Average number of Persons Involved in providing information* ( $\overline{PI}$ ) is 2.17 persons per enterprise.

Size class	1(b)	2	3	4	5	6+	Average number of Per- sons Involved in providing information
10-19	47.40	37.64	11.66	2.25	0.65	0.41	1.73
20-49	38.79	37.30	16.65	4.44	1.94	0.87	1.98
50-99	30.46	36.46	19.04	7.15	4.49	2.40	2.30
100-249	20.15	34.04	24.67	9.98	6.20	4.96	2.71
250+	11.40	27.86	24.84	12.19	10.43	13.28	3.53
Total	36.51	35.82	16.44	5.26	3.13	2.83	2.17

Table 5.1.3 - Respondents ICT2015: number of persons Involved in providing information by size class (percentages)(a)

Source: Istat - ICT survey: study on the burden measurement system

(a) Missing: 686.

(b) In this category enterprises answered 'zero persons involved' are included.

Using central value of each time classes to retrieve information (even if the classes do not have the same width), it is possible to calculate the *Average Retrieval Time* ( $\overline{\text{ART}}$ ) as reported in Table 5.1.4. Standard deviation shows high dispersion of the data around the mean.

	Table 5.1.4 - Average	<b>Retrieval Time</b>	by size	class /	(a)
--	-----------------------	-----------------------	---------	---------	-----

Size class	Missing	Ν	Mean <i>(minutes)</i>	Standard deviation
10-19	97	9,026	53.29	83.19
20-49	42	3,624	59.73	88.62
50-99	15	1,959	74.07	114.71
100-249	26	2,061	90.74	139.07
250+	30	2,438	125.86	182.58
Total	210	19,108	69.94	114.20

Source: Istat - ICT survey: study on the burden measurement system

(a) Central values of first 5 time classes shown in Table 5.1.2 have been used.

Paradata are also useful in measuring respondent burden in terms of time spent to complete the questionnaire, so that it is possible to calculate the *Average Compilation Time* ( $\overline{ACT}$ ) (2015, Masselli *et al.*; 2014, Nuccitelli *et al.*). In Table 5.1.5 standard deviation shows high dispersion of data and an average time of about 47 minutes increasing passing from small enterprises (42 min) to large ones (65 min). Data show a certain direct relation between size classes and net time needed to complete the questionnaire.

Size class	Ν	Mean ( <i>minutes</i> )	Standard deviation
10-19	8,962	41.95	50.75
20-49	3,614	43.30	56.64
50-99	1,952	46.47	64.40
100-249	2,064	54.11	75.37
250+	2,465	64.68	95.44
Total	19,057	46.92	64.01

Table 5.1.5 - Average Compilation Time in minutes by size class

Source: Istat - ICT survey: study on the burden measurement system

In Tables 5.1.6 and 5.1.7 are presented results from a question included in the *Burden section* only during the first wave of the survey hosted by the new data collection system (2014). The question was about difficulties encountered by respondents in filling in the questionnaire. It was asked about *IT difficulties* caused by the new system, *conceptual difficulties* related to lack of clarity or of supporting information, *ease of data availability* and on length of the questionnaire. About 1 out of 2 respondents was experiencing no difficulties (49%). Later on, we tried to solve the problem of conceptual difficulties improving language used in the questions, using more effective FAQ and instruction for filling questionnaire uploaded in web site of survey and adding more tooltips in the web model.

		Respondent to ICT and not to difficulties	Respondent both to ICT and difficulties
	Total -	Percen	tages
10-19	1,831	5.74	94.26
20-49	9,021	6.52	93.48
50-99	3,329	7.08	92.92
100-249	2,308	9.56	90.44
250+	2,061	10.49	89.51
Total	18,550	7.00	93.00

#### Table 5.1.6 - Respondents to ICT2014 survey and to difficulties question, by size class (absolute value and percentages)

Source: Istat - ICT survey: study on the burden measurement system

As it is shown in Table 5.1.7, 1 enterprise out of 3 indicates not more than 2 difficulties encountered in compiling the questionnaire.

Table 5.1.7 - Respondents ICT2014 by number of difficulties encountered in compiling
the questionnaire, by size class (percentages)

	Number of difficulties							
Size class	0/missing	1	2	3	4+			
10-19	50.13	21.44	14.75	7.87	5.81			
20-49	56.32	20.04	13.25	6.58	3.81			
50-99	60.80	19.21	11.89	5.77	2.33			
100-249	65.05	18.79	10.54	3.39	2.24			
250+	68.93	17.76	8.45	3.16	1.69			
Total	56.24	20.22	12.96	6.38	4.20			
Total	56.24	20.22	12.96	6.38	4.20			

Source: Istat - ICT survey: study on the burden measurement system

In general, for all sizes of enterprise the highest difficulty was related to the length of the questionnaire (21%), also conceptual difficulties were big obstacles to fill in the questionnaire (26,5%); finally, very few indicated difficulties encountered in the new data collection tool (9%) (Table 5.1.8).

Size class	Technical and IT problems	Difficulties in navigating through the questionnaire sections	Too many que- stions	Insufficiently clear que- stions	Unclear or not exhaus- tive answer options	Not enough information supporting the understanding of questions	Excessive distance between in- formation re- quested and information available in the business	No difficulties encountered
10-19	9.33	1.52	23.74	22.30	11.66	17.83	14.54	44.39
20-49	8.44	1.20	20.46	17.87	9.28	13.97	11.93	49.80
50-99	7.57	0.92	18.78	13.29	8.01	10.29	11.60	53.71
100-249	5.57	1.47	17.86	10.32	7.97	7.65	9.34	55.49
250+	3.94	0.95	15.86	8.80	7.50	5.20	9.36	58.45
Total	7.94	1.32	21.04	17.64	9.95	13.72	12.59	49.24

#### Table 5.1.8 - Respondents ICT2014 by type of difficulties encountered in compiling the questionnaire, by size class (percentages)

Source: Istat - ICT survey: study on the burden measurement system

#### Figure 5.1 reports the summary report on response burden for ICT survey.

#### Figure 5.1 - Summary report for ICT survey year 2015

(1)	<b>ACT</b> = Average Compilation Time in minutes	: 46.92							
(2)	<b>ART</b> = Average Retrieval Time : 69.94								
(3*)	$\overline{\mathbf{PI}}$ = Average number of Persons Involved in p	roviding infor	mation:	2.17					
(4*)	Frequency distribution of enterprises per numb	er of problems	s encour	ntered					
Numl	er of difficulties declared by the respondent	0/missing	1	2	3	4	5	6	7
	% of respondent enterprises	56.24	20.22	12.96	6.38	2.73	1.04	0.23	0.20
	Number of respondent enterprises	10,432	3,751	2,405	1,183	506	192	43	38
	- Number of respondent businesses used <sup>7</sup> :	19	9,421					•	
	- Coordination function used in selection of the sample: negative coordination								
*Note th	hat for indicators 3 and 4 data from ICT year 201	4 are used							

Source: Istat - ICT survey: study on the burden measurement system

#### 5.2 First results of burden indicators on LES survey

Since the '80s of the last century, Istat has conducted the monthly survey on employment, working hours, wages and labour costs in large enterprises (LES). Starting from 2012 a section about job vacancy has been included in the questionnaire at the end of every quarter (in March, June, September and December). This survey on the *large enterprises* contributes along with other two quarterly surveys (one concerns job vacancy and hours worked – Vela, the other one concerns Employment Remuneration and Social Security Contributions – Oros) to determine indicators on the input and labour costs in all enterprises with employees. The values obtained through the integration of these three different statistical surveys are sent to Eurostat in compliance with the following European regulations:

- 1. STS-term statistics (no. 1165/98) and subsequent amendments and additions to the number of persons employed, hours worked and gross and salary wages.
- 2. Labour Cost Index (n. 450/2003).
- 3. Regulation on job vacancies of the European Parliament and of the Council (no. 453/2008) and on the Commission's implementing regulations (no. 1062/2008 and no. 19/2009).

The LES survey refers to enterprises with more than an average of 500 employees at the base year  $2010^8$ . These are about 1,250 and are monitored heavily to minimise the non-response rate, that has been registered to be about 17% for preliminary estimates and only  $3.5\%^9$  for the final estimates released in April 2016.

It is worth reporting that, in order to renew the panel for the base year 2015, a set of 380 new enterprises started to be surveyed. As these enterprises do not contribute to the current results, they are monitored more lightly: at April 2016 the non-response rate on this set of units has been registered to be about 40%.

<sup>8</sup> This base refers to 2015 year of survey, but it changes periodically.

<sup>9</sup> Monthly reminders (by e-mail and fax) and intensive follow-ups by phone are addressed to not responding LES units. In 2015 once a year a warning with penalty (registered letter with return receipt) was sent to firms that had not answered to LES for two or more months.

In 2015 this survey started to be managed with the new data collection system integrated in the Business Statistical Portal and the questionnaire was developed in GX system.

As ICT survey, with the migration to the Business Statistical Portal some significant innovations were introduced. They regarded:

- number of questions: some added and others removed from previously 2014 edition;
- overall design of the questionnaire;
- the way inconsistencies are highlighted in the web form when rules violations are triggered.

Furthermore, the already mentioned *Burden section* was added in the questionnaire (section K). It's important to underline that answering to this section was not compulsory and no alerts appeared on the screen if the section was not filled.

In the following, results of the analysis on both the compilation time and the *Burden section* are presented.

As already said, response rate of LES survey in 2015 is different depending on whether the enterprises belong to panel of the base 2010 (1,250 enterprises) or to the renewed panel for new base 2015 (not already considered for published data, 380 units).

In addition, as only the subset of businesses who provided data for 12 months is considered to calculate the burden indicators (see Paragraph 4.1), the response rate for 2015 is 88% (1,422 enterprises as respondents over 1,630 enterprises in total).

The Average Compilation Time ( $\overline{ACT}$ ) is almost 43 minutes, more or less the same time as standard deviation.

#### Table 5.2.1 - Average Compilation Time in minutes

	Ν	Mean ( <i>minutes</i> )	Standard deviation
Total	1,422	42.6	43.7

Source: Istat - LES survey: study on the burden measurement system

The respondents to section K of questionnaire of December were 997, while the non respondent units were 362. In this case the number of units that were supposed to respond were 1,765: in fact this section is referred to each respondent unit (KAU) (Table 5.2.2).

#### Table 5.2.2 - Respondents LES 2015: Burden section

	Number	Respondent to LES December	Respondent both to LES December
	of units	and not to burden	and burden
Total	1,765	362	997

Source: Istat - LES survey: study on the burden measurement system

Table 5.2.3 shows the time to retrieve information by size classes. As we see, nearly half of respondents needed more than 1 hour and up to 3 hours, while the 31% needed more than half an hour and up to 1 hour to fill the questionnaire.

### Table 5.2.3 - Respondents LES2015: time to retrieve information by size class (percentages)(a)

	Time to retrieve information						
-	Up to 30 minutes	More than half an hour and up to 1 hour	More than 1 hour and up to 3 hours	More than 3 hours and up to 10 hours	More than 10 hours and up to 30 hours		
Total	12.1	31.0	44.4	12.0	0.4		

Source: Istat - LES survey: study on the burden measurement system (a) Missing: 367.

The Average number of Persons Involved in providing information ( $\overline{PI}$ ) is 1,5 persons per enterprise (Table 5.2.4).

Table 5.2.4 - Respondents LES2015: number of persons Involved in providing information by size class (percentages) (a)

	1 <sup>(b)</sup>	2	3	4	5	6+	Average number of Persons Involved in providing information
Total	64.1	24.0	9.4	2.5	0	0	1.5

Source: Istat - LES survey: study on the burden measurement system (a) Missing: 367.

(b) In this category enterprises answered 'zero persons involved' are included.

Finally the *Average Retrieval Time* ( $\overline{\text{ART}}$ ) is reported in Table 5.2.5. Standard deviation shows high dispersion of the data around the mean.

Table 5.2.5 - Average Retrieval Time

	Missing	Ν	Mean <i>(minutes)</i>	Standard deviation
Total	367	997	120.8	128.7

Source: Istat - LES survey: study on the burden measurement system

As it is shown in Table 5.2.6, less 10% of enterprises encountered 2 or more difficulties compiling the questionnaire.

## Table 5.2.6 - Respondents LES 2015 by number of difficulties encountered in compiling the questionnaire (percentages)

	Number of difficulties						
	0/missing	1	2	3	4+		
Total	67.03	23.27	7.56	1.69	0.44		

Source: Istat - LES survey: study on the burden measurement system

The summary report on response burden for LES survey is shown in Figure 5.2.

#### Figure 5.2 - Summary report for LES survey year 2015

(1)	<b>ACT</b> = Average Compilation Time in minutes: 42.6					
(2)	<b>ART</b> = Average Retrieval Time : 120.8					
(3)	$\mathbf{\overline{PI}}$ = Average number of Persons Involved in providing in	nformation:1.5				
(4)	Frequency distribution of enterprises per number of probl	lems encountered	d			
	Number of difficulties declared by the respondent		0/missing	1	2 3	4+
	Number of difficulties declared by the respondent % of respo	ndent enterprises	<i>0/missing</i> 67.03	<i>1</i> 23.27	<b>2 3</b> 7.56 1.69	<i>4</i> + 0.44
	Number of difficulties declared by the respondent % of respon Number of respon	ndent enterprises ndent enterprises	0/missing 67.03 911	1 23.27 316	<b>2 3</b> 7.56 1.69 103 23	<u>4+</u> 0.44 6
	Number of difficulties declared by the respondent % of respon Number of respon - Number of respondent businesses used:	ndent enterprises ndent enterprises 1,422	<i>0/missing</i> 67.03 911	1 23.27 316	<b>2 3</b> 7.56 1.69 103 23	<u>4+</u> 0.44 6

Source: Istat - LES survey: study on the burden measurement system

#### 5.3 First results of burden indicators on Industrial Turnover and new Orders survey

The monthly turnover index is designed to measure the performance of industrial sales over time, limited to mining and manufacturing activities. The orders index measures the dynamic of the value of new orders received by enterprises from clients each month. This second indicator is based on the information provided only by a sub-group of enterprises participating in the survey, in particular by the industrial sectors which usually work on commissioned orders.

The reporting unit for both variables is the enterprise; however, if a firm's turnover/orders refers to more than one economic activity (at three-digit level of NACE), data are collected separately for each kind of activity unit (KAU).

Turnover is defined as the total value of all the invoices issued during the month, for sales in the domestic or non-domestic market (divided into Euro and non-Euro areas), net of VAT invoiced to clients and any discounts or rebates shown in the invoice, before expenses (shipping, packaging, etc.) or other duties.

The Industrial Turnover Indices are sent to Eurostat in compliance with the European regulation no. 1165/98 on STS-term statistics and subsequent amendments and additions.

Orders include all the new orders – in term of value – received and accepted during the month. The information is disaggregated according to whether the orders come from domestic or non-domestic clients.

The survey refers to enterprises with more than 20 employees<sup>10</sup>, the sample is selected from the Statistical Business Register with a cut off criteria<sup>11</sup>, the index is a fixed index with 2010 as base year<sup>12</sup>. The sample consists of about 8,900 companies that are monitored carefully to ensure a low rate of non response, that was near to 10% for preliminary monthly estimates and 4%<sup>13</sup> for the final estimates released in November.

<sup>10</sup> For particular sectors characterised by small enterprises the size could be lower.

<sup>11</sup> The sample is extracted to cover for all sectors -defined at 3 digit level of NACE - more or less the 70% in terms of turnover of each sector.

<sup>12</sup> See note n.6

<sup>13</sup> Monthly reminders by e-mail and intensive follow-ups by phone are made to non-respondent units. A legal annual warning (with a financial penalty) is sent to firms that didn't answered for at least two months in a year.

At the beginning of 2016, a set of 2,000 enterprises has been added to the sample to prepare the next rebasing (base year=2015); as these enterprises are 'new' and not very skilled with the survey, the rate of non response of this particular subset of enterprises was higher<sup>14</sup>.

In 2015 the survey was redesigned as a Computer Assisted Web Interviewing (CAWI) survey and the questionnaire is available through the new data collection system integrated into the Business Statistical Portal (GX).

As for the other two surveys previously analysed, with the introduction of the Business Statistical Portal some significant innovations were introduced:

- reduction of number of questions: the split of foreign orders into euro and non-euro area was removed;
- introduction of some automatic checks to highlight inconsistencies during the filling of the electronic form.

Also the turnover questionnaire has a new section to measure *respondent burden*. The filling of this section was not mandatory for the 2015 and no alert appeared if the section was not filled.

In the following, the results of the analysis on the compilation time and on the *Burden section* are presented.

As seen previously, survey's response rate in 2015 is different according to the different purposes of the analysis: if the enterprises belong to the sample referred to the 2010 base (6,500 enterprises) or to the new set of enterprises extracted for the 2015 rebasing and not already considered in the calculation of the index currently disseminated.

In addition, as only the subset of businesses which provided data for 12 months is considered to calculate the burden indicators (see Paragraph 4.1), the response rate for 2015 is 90% (6,252 respondents out of 6,928 sampling enterprises).

The Average Compilation Time ( $\overline{ACT}$ ) is equal to 5.4 minutes, the standard deviation to 6.2.

<sup>14</sup> These enterprises don't contribute to the index calculation before the introduction of 2015 as base year.
#### Table 5.3.1 - Compilation time in minutes - ACT

	Ν	Mean <i>(minutes)</i>	Standard deviation
Total	6,252	5.4	6.2

Source: Istat - Turnover and new Orders survey: study on the burden measurement system

The respondents to the *Burden section* in December were 1,468, while the non respondent units were 5,059. In this case the number of units that were supposed to respond were 6,527 because this section is referred to each respondent unit (KAU) (Table 5.3.2).

#### Table 5.3.2 - Respondents Turnover 2015: Burden section

Number F		Respondent to LES December	Respondent both to LES December
of units		and not to burden	and burden
Total	6,527	5,059	1,468

Source: Istat - Turnover and new Orders survey: study on the burden measurement system

Table 5.3.3 shows the time to retrieve information by size classes; the 66.8% of respondents needs less than half an hour, while the 24.5% needs more than half an hour but less than 1 hour.

# Table 5.3.3 - Respondents Turnover 2015: time to retrieve information by size class (percentages)(a)

		Time to retrieve information						
	Up to 30 minutes	More than half an hour and up to 1 hour	More than 1 hou					
Total	66.8	24.5	8.8					

Source: Istat - Turnover and new Orders survey: study on the burden measurement system (a) Missing=5,059.

The Average number of Persons Involved in providing information ( $\overline{PI}$ ) is 1.3 persons per enterprise (Table 5.3.4), the 95.5% of enterprises involving a single person in filling in the questionnaire.

	inion	nation	percenta	ages)			
	1 <sup>(b)</sup>	2	3	4	5	6+	Average number of Persons Involved in providing information
Total	95.5	3.8	0.6	0.1	0	0	1.3

# Table 5.3.4 - Respondents Turnover 2015: number of persons Involved in providing information (percentages)

Source: Istat - Turnover and new Orders survey: study on the burden measurement system (a) Missing: 5,059.

(b) In this category enterprises answered 'zero persons involved' are included.

Finally the *Average Retrieval Time* ( $\overline{\text{ART}}$ ) is reported in Table 5.3.5. Standard deviation shows very low dispersion of the data around the mean.

#### Table 5.3.5 - Time to retrieve information (a)

	Missing	N	Mean <i>(minutes)</i>	Standard deviation
Total	5,059	1,468	31.6	30.25

Source: Istat - Turnover and new Orders survey: study on the burden measurement system (a)Central values of time classes shown in Table 1 have been used.

As it is shown in Table 5.3.6, 1 less than 2% of enterprises encountered 2 or more difficulties compiling the questionnaire.

# Table 5.3.6 - Respondents Turnover survey by number of difficulties encountered in compiling the questionnaire (percentages)

	Number of difficulties								
	0/missing	1	2	3	4+				
Total	81.85	19.27	0.31	0.05	0.05				

Source: Istat - Turnover and new Orders survey: study on the burden measurement system

The summary report on response burden for Turnover survey is shown below.

(1)	<b>ACT</b> = Average Compilation Time in minutes: 5.4						
(2)	$\overrightarrow{\textbf{ART}} = \text{Average Retrieval Time}: 31.6$						
(3)	$\mathbf{\overline{P}I}$ = Average number of Persons Involved in providing information	nation:1.3					
(4)	Frequency distribution of enterprises per number of problems en	encounter	ed				
Numb	per of difficulties declared by the respondent		0/missing	1	2	3	4+
Numb	ber of difficulties declared by the respondent % of respondent ent	nterprises	<i>0/missing</i> 81.85	<i>1</i> 19.27	2 0.31	<b>3</b> 0.05	<i>4</i> + 0.05
Numb	ber of difficulties declared by the respondent % of respondent ent Number of respondent ent	nterprises	<i>0/missing</i> 81.85 5,343	<i>1</i> 19.27 1,258	2 0.31 20	3 0.05 3	<u>4+</u> 0.05 3
Numb	ber of difficulties declared by the respondent % of respondent ent Number of respondent businesses used: 6,252	nterprises nterprises 2	0/missing 81.85 5,343	<i>1</i> 19.27 1,258	2 0.31 20	3 0.05 3	<u>4+</u> 0.05 3

#### Figure 5.3 - Summary report for Turnover survey year 2015

Source: Istat - Turnover and new Orders survey: study on the burden measurement system

# 6. Automatic Procedure for Burden Indicators by Survey (BBS)

A generalised software application for the production of the indicators described in the previous section has been implemented.

This procedure runs for all the surveys in the Business Portal architecture, using the GX software for the electronic questionnaire (the possibility of expanding it also for questionnaires developed with other software systems is under study).

In practice, each time a new survey starts, the responsible of the survey will define:

- the type of the *Burden section* to be used for his survey: as described in Section 4, the general structure of this section is customised according the characteristics of the survey questionnaires (short, medium or long);
- the use of sampling coordination function;
- the reference of RDBMS where the *Burden section* microdata are stored;
- the year of reference of SBR ASIA archive to retrieve information regarding size class of employees or Economic Activity sector of the businesses.

Further information to be provided by the responsible of the survey is the starting and ending date of the survey to be considered, in order to select the corresponding paradata to be stored in the RDBMS.

In the Business Statistical Portal Section 'Online Report', the procedure provides a new report to be selected in the list of those which can be generated: Summary report on response burden.

31175 - Rilevazione sulle	tecnologie dell'i	nformazio	ne e della co	municazione	e nelle imp	orese (ICT	7) 2015	ls
e Impresa * Rilevazione * Questionari *	Reportistica 💌 U	Jtilità 🔻 /	Amministrazior	е т				
Response Burg	den calcula	ted on	6/12/20	16 with cl	osed su	rvev		
Average Compilation Time	46 92	,	Average re	trieval Tim	e	/		69.94
Average compliation rime	providing info	rmation						2.17
Frequency distribution of enterprises pe	r number of p	oroblem	encounter	ed				۲,17
Number of difficulties declared by the	0 / Missing	I	2	3	4	5	6	7+
respondent								
% of respondent enterprises	56,24	20,22	12,96	6,38	2,73	1,04	0,23	0,20
Number of respondent enterprises	10432	3751	2405	1183	506	192	43	38
Number of respondent business used								19421
Number of unit respondent business us	ed							
Coordination function used in selection	of the sampl	е			_			Negative
						C	Close	

#### Figure 6.1 - Response burden report for the ICT survey

The number of columns regarding 'Number of difficulties declared by the respondent' will be different depending of the type of the *Burden section*.

During the data collection phase, the person in charge can run the reports and the software application processes both the paradata and the results obtained on the *Burden section* to generate the standard burden indicators by survey. However, the results will be stored in RDBMS only when the survey is definitively closed (the person in charge will close the calculation with the appropriate button).

The report produces also results by size class of employees, using the information already stored.

e Impresa * Rilevazione * Questionari * Re	portistica * U	tilità 💌 Amminis	trazione 💌			
Response Burden calculated on 10	6/12/2016	with closed s	survey S	iize class	10-19 20-49 50-99 100-249 • 250+	
Average Compilation Time	64,68	Averag	ge retrieval T	ime		125,86
Average number of Persons Involved in pr	oviding infor	mation				3,53
Frequency distribution of enterprises per r	number of p	roblem encou	intered			
Number of difficulties declared by the respondent	0 / Missing	1	2	3	4+	
% of respondent enterprises	68,93	17,76	8,45	3,16	1,69	
Number of respondent enterprises	172 <del>9</del>	445	212	79	42	
Number of respondent business used	1					2509
reamber of unit respondent business used						News

#### Figure 6.2 - Response burden report for the ICT survey, by size class of employees

## 7. Burden indicators in terms of persistency (BBB)

The persistency indicators aim at measuring burden from the point of view of a single enterprise potentially involved in several Istat surveys in a given time period. The main reason why Istat has decided to calculate the persistency indicators is that multiple requests addressed to the same enterprise may reduce the quality of the response given.

In the last years, many National Statistical Institutes have been increasingly using sampling coordination techniques to reduce the overlap between samples of different surveys. Nevertheless, large enterprises are more likely to be selected into survey samples, causing a 'persistent' burden to these statistical units. Indeed, they are often selected with probability 1 from different surveys or from the same survey in consecutive time periods.

In this report we produce two groups of persistency indicators referring to planned and actual burden for year 2015. The first group of indicators is based on the number of surveys a unit is selected for, while the second one considers the number of questionnaires the enterprise fills in. The calculation of the persistency indicators does not imply further burden on the enterprises, because it is based on information already available in Istat databases.

The indicators we present in this report are based on the following quantities:

S1, S2, ..., Sn = number of enterprises selected into (exactly) 1, 2, ..., n surveys during year t

*R1, R2, ..., Rn* = number of enterprises that have filled in (exactly) the questionnaires of 1, 2, ..., *n* surveys during year *t*.

From the quantities above we have calculated the following indicators:

 $S'_{k} = \frac{\sum_{i=k}^{n} S_{i}}{N} \cdot 100 = \text{percentage of active enterprises selected into at least } k$ surveys during year t

 $R'_{k} = \frac{\sum_{i=k}^{m} R_{i}}{N} \cdot 100 = \text{percentage of active enterprises responding to at least } k$ surveys during year *t*, where N is the total number of active enterprises according to Asia, the Italian Statistical Business Register (SBR)<sup>15</sup>, n and M are the maximum number of surveys an enterprise is involved in during year  $t^{16}$ .

Table 7.1 - Enterprises selected into at least k surveys during year 2015

Number of surveys (k)	1	2	3	4	5	6	7
% active enterprises selected into k surveys	4.96	1.20	0.56	0.21	0.32	0.14	0.09

Source: Istat - Study on the burden measurement system

#### Table 7.2 - Enterprises responding to at least k surveys during year 2015

Number of surveys (k)	1	2	3	4	5	6	7
% active enterprises responding to k surveys	2.37	0.66	0.32	0.19	0.12	0.08	0.05

Source: Istat - Study on the burden measurement system

The indicators of persistency have been finally calculated by size classes (in terms of number of persons employed) and economic activity, exploiting the information available from the SBR Asia<sup>17</sup>.

Size classes	0 - 9	10 - 19	20 - 49	50 -249	≥250	Total
% active enterprises selected into surveys	2.49	42.77	67.34	93.33	100	4.960

Source: Istat - Study on the burden measurement system

#### Table 7.4 - Enterprises responding into surveys during year 2015, by size classes

Size classes	0 - 9	10 - 19	20 - 49	50 -249	≥250	Total
% active enterprises responding to surveys	0.82	22.31	43.43	72.67	96.07	2.37

Source: Istat - Study on the burden measurement system

<sup>15</sup> Calculation is based on the updated SBR Asia from which the sampling frame has been taken (2013).

<sup>16</sup> *n* and *m* are the maximum number of surveys an enterprises involved, in the sense of planned and actual burden, respectively.

<sup>17</sup> About SBR Asia year of reference, see footnote above.

Table 7.3 shows that the share of enterprises that are selected into several surveys rapidly increases with the number of persons employed. In particular, it emerged that 2.5% of microenterprises (<10 persons employed) are involved in surveys, while this percentage grows to 49.5% for small enterprises (10-49 p. e.) and 94.5% for medium and large enterprises ( $\geq$ 50 p. e.). The share is 100% for large enterprises ( $\geq$ 250 p. e.) because in most Istat business surveys they are enumerated. We can also observe from Table 7.4 that unit response rate of larger enterprises is very high.

Table 7.5 - Enterprises selected into surveys during year 2015, by economic activity

Economic activity	Industry	Construction	Services	Total
% active enterprises selected into surveys	15.30	3.36	3.86	4.96

Source: Istat - Study on the burden measurement system

#### Table 7.6 - Enterprises responding to surveys during year 2015, by economic activity

Economic activity	Industry	Construction	Services	Total
% active enterprises responding to surveys	8.29	1.54	1.74	2.37

Source: Istat - Study on the burden measurement system

Tables 7.5 and 7.6 finally show that Industry has the largest share of enterprises selected and responding to Istat surveys.

## 8. Conclusions and perspectives

The activities carried out in 2016 by the Istat Working Group on the measurement of burden for surveys on businesses have been focussed on the development and testing of a set of standard indicators (and metadata) on burden to be calculated in a systematic way.

So far, the indicators 'by survey' have been tested on three business surveys with different characteristics ('ICT survey', 'Monthly survey on employment and labour cost in large enterprises' and 'Industrial turnover and orders') and the results are reported in Section 5.

The test has been useful not only to assess the proposed indicators, but also to define the requirements for the software application that should automatically calculate them. The proposed strategy is to add a standard *Burden section* to all surveys that migrate to the business portal IT environment, so as to implement a generalised software application that process both the paradata and the responses collected in the *Burden section* and calculate the standard burden indicators by survey (BBS). An example of this procedure has already been implemented for the GX data collection system, as reported in Section 6.

In perspective, it is planned that the burden indicators by survey will be stored in the Istat official system for the documentation of reference metadata and quality indicators, named SIDI/SIQual. This database should automatically be updated with these indicators in order to disseminate them to users through their integration into the National Quality Reports<sup>18</sup> that are produced through SIDI/SIQual and disseminated on the website starting from June 2018. These enhancements are still in progress as SIDI/SIQual system is also starting to be redesigned.

Furthermore, the Working group tested the Burden indicators in terms of persistency for year 2015 and the results are reported in Section 7. A software procedure should be developed to calculate annually and automatically also this set of indicators. In this case the source to derive the indicators will be the db built to manage the outcome of data collection of all the businesses surveys.

<sup>18 &</sup>quot;Schede standard di qualità" in Italian are available at: <u>https://www.istat.it/it/metodi-e-strumenti/strumenti-per-la-qualit%C3%A0/schede-standard-di-qualit%C3%A0.</u>

The last task of the Working Group on the measurement of burden for surveys on businesses was to make suggestions on the indicators interpretation in order to identify strategies for reducing the burden and hopefully improving the quality of data produced by the surveys.

From the test carried out, it resulted quite clear that the specific expertise of the survey managers is necessary to interpret the indicators and identify the possible areas of improvement. The survey managers also know, for example, if there are alternative sources (*e.g.* administrative data, 'big data', web scraping, etc.) that can be exploited to reduce the burden. For example, following this study on the ICT survey, some reflections were made about the need to reduce respond burden not only acting more on development of the questions and of their explanation inside the questionnaire, but also trying to experiment other data source to have information similar to collected ones by survey. In fact in year 2017 web scraping and machine learning techniques were used to produce alternative experimental estimates on three variables collected by ICT survey (enterprises offering in their web sites web ordering, job application functionality and link to social media).

However, often the implementation of improvement actions implies the involvement and coordination of different surveys. This is the case, obviously, of the application of negative coordination of samples among different surveys in order to reduce the burden in terms of persistency.

Another example is the reduction of redundancies in information asked on different surveys: with the introduction of the Istat Business Statistical Portal the demographic information on the enterprises are now managed and updated directly by the Portal and not asked separately by all the surveys<sup>19</sup>, other information could be managed in the same way.

It is also well known that the 'subjective' burden can be reduced improving the data collection tool. Some examples of functionalities that are deemed useful and are not implemented at the moment are the possibility to print the draft version of the questionnaire, or the consultation of questionnaires of previous editions.

<sup>19</sup> In the Business Portal environment, each respondent, before filling in a survey questionnaire, can check and update, if necessary, enterprise' stored demographic data. In this way, this information is available for all the surveys where enterprise is involved in.

Another functionality which could be improved in the Business Portal concerns the feedback of statistical data to the businesses. In particular, relevant data for the business are going to be provided together with some functionalities connected to them, like the elaboration of 'Graphical presentation' according to the business features (economic sector, dimension and territorial location).

In conclusion, this work has to be analysed taking into account the transitional period during which it has been implemented. Major changes occurred to the organisational architecture of the Institute, so that implementation of such a system of indicators and the proposal for a full dissemination should be tailored to the new scenario. Nevertheless, the underlying concepts, both from a methodological and IT points of view, have been defined according to the generalised criteria as requested from the modernisation process all the NSI are going through.

Hence the proposed system can represent anyway a milestone and the cost for its adaptation so as to use it systematically should be low enough.

## References

Couper, M.P. 1998. "Measuring survey quality in a CASIC environment". In *Proceedings of the Survey Research Methods Section*, American Statistical Association: 41-49.

Edwards, W.S., and D. Cantor. 1991. "Towards a Response Model in Establishment Surveys". In Biemer, P.P., R.M. Groves, L. Lyberg, N. Mathiowetz, and S. Sudman (Eds.). *Measurement Errors in Surveys*. Hoboken, NJ, U.S.: Wiley & Sons.

European Commission, Eurostat, Directors of Methodology – DIME. 2014. "Memobust Handbook - Handbook on Methodology of Modern Business Statistics". *Methodological handbooks*. Collaboration in Research and Methodology for Official Statistics – CROS: Eurostat.

Fazio, N., M. Murgia, and A. Nunnari. 2013. "The Business Statistical Portal: a new way of organising and managing data collection process for business surveys in Istat". In *Proceedings of the UNECE Seminar on Data Collection*. Geneva, Switzerland, 25<sup>th</sup> - 27<sup>th</sup> September 2013.

Italian National Institute of Statistics - Istat. 2016. *Istat's Modernisation Programme*. Roma: Istat.

https://www.istat.it/it/files//2011/04/IstatsModernistionProgramme\_EN.pdf.

Masselli, M., A. Nuccitelli, and A. Laureti Palma. 2018. "The behaviour of respondents while filling in a web questionnaire: the case of the Italian Business R&D Survey. *Istat working papers*, n. 24/2016. Roma, Italy: Istat.

Nuccitelli, A., A. Nunnari, A. Nurra, and N.R. Fazio. 2014. "Towards a systematic use of paradata: the case of the Italian ICT survey". Presentation at the 7<sup>th</sup> Internet Survey Methodology Workshop. Free University of Bozen-Bolzano, Italy.

Sudman, S., D.K. Willimack, E. Nichols, and T.L. Mesenbourg. 2000. "Exploratory Research at the U.S. Census Bureau on the Survey Response Process in Large Companies". In Proceedings of the *Second International Conference on Establishment Surveys*. American Statistical Association: 327-337. Willimack, D.K., and E. Nichols. 2001. "Building an Alternative Response Process Model for Business Surveys". In Proceedings of the *Annual Meeting of the American Statistical Association*.

# Handling Multinational Enterprises: Istat's early experience in Large Cases Unit

Simone Ambroselli, Sonia Amante, Elisabetta Bilotta, Barbara Gentili, Barbara Iaconelli, Luigia Mattei, Chiara Orsini, Fabiana Sartor, Ilaria Straccamore<sup>1</sup>

# Abstract

The way Multinational Enterprises (MNEs) organise and reorganise their businesses worldwide has an impact on the production of the official statistics. During 2016, a dedicated team specifically devoted to large and complex units was established within the Directorate for Economic Statistics of the Italian National Institute of Statistics - Istat aiming at identifying the effects of complex restructuring at the early stages of the statistical production process. The main objective of this paper is describing the results of several analyses the Large Cases Unit (LCU) Team has conducted during the first phase of the project in evaluating size, significance and complexity aspects according to both the characteristics of national economy and the internal organisation of Istat.

**Keywords:** Multinational Enterprises (MNEs), Large Cases Unit (LCU), official statistics, globalisation.

<sup>1</sup> Simone Ambroselli (ambrosel@istat.it); Sonia Amante (amante@istat.it); Elisabetta Bilotta (bilotta@istat.it); Barbara Gentili (bagentil@istat.it); Barbara Iaconelli (iaconell@istat.it); Luigia Mattei (mattei@istat.it); Chiara Orsini (chorsini@istat.it); Fabiana Sartor (sartor@istat.it); Ilaria Straccamore (straccamore@istat.it), Italian National Institute of Statistics - Istat.

The views and opinions expressed are those of the authors and do not necessarily reflect the official policy or position of the Italian National Institute of Statistics - Istat.

The authors would like to thank the anonymous reviewers for their comments and suggestions, which enhanced the quality of this article.

# 1. Introduction<sup>2</sup>

The way Multinational Enterprise Groups (from hereon will be referred to as MNEs) organise and reorganise their businesses worldwide has an impact on the production of the official statistics. Complex restructuring may affect single legal entities belonging to the MNEs, every kind of statistical unit used in the statistical production processes, all the National Statistics domains and the International comparability. For official statistics, globalisation is "one of the largest 'statistical measurement' challenges facing producers of economic statistics today" (Hussain *et al.*, 2019: 44). In fact, according to Sturgeon (2013), globalisation is a cross-border process for which an integrated and international standard approach mainly based on a better use of existing data is essential for handling the issue.

The Italian National Institute of Statistics - Istat during the last ten years has been involved in several international activities in order to manage the implications deriving from globalisation, such as the EuroGroup Register (EGR)<sup>3</sup>, International profiling activities<sup>4</sup>, Early Warning System, Gross National Income (GNI) MNE Pilot, Foreign Affiliates Statistics (FATS), International Sources statistics, and various projects aimed at improving the quality of National Accounts, concerning both input and output data. Eurostat has played a central role both in the research of a standard approach and in sharing common tools and practices, but also other international institutions have been increasingly interested in developing new ways to afford the problem in an international perspective. The United Nations Economic Commission for Europe (UNECE), Eurostat and the Organisation for Economic Co-operation and Development (OECD) have published in 2011 a guide on the Impact

<sup>2</sup> Although the article is the result of a joint work, the single parts are authored as follows: Sections 1, 2, 3, and 6 by Simone Ambroselli; Paragraph 4.1 by Barbara Gentili; Paragraph 4.2 by Fabiana Sartor; Paragraph 4.3 by Barbara Iaconelli; Paragraph 4.4 by Ilaria Straccamore; Paragraph 4.5 by Luigia Mattei; Paragraph 4.6 by Sonia Amante; Paragraph 5.1 by Chiara Orsini; Paragraph 5.2 by Elisabetta Bilotta. Proofreading by Fabiana Sartor.

<sup>3</sup> EuroGroups Register (EGR) is the statistical business register of multinational enterprise groups having at least one legal unit in the territory of the EU or EFTA countries.

https://ec.europa.eu/eurostat/web/structural-business-statistics/structural-business-statistics/eurogroups-register
UNECE, Guidelines on Statistical Business Registers (2015): "6.111 Profiling is the practice of using company accounts, often accompanied by interviews with senior enterprise officials, to build and define the structure of enterprises, mainly those involved in large complex enterprise groups. The resulting profiles are used to produce a reporting structure appropriate for the surveys conducted by the NSI. Profiling usually involves establishing contact with the enterprise being profiled to develop a good understanding of its structure. It is possible, however, to complete smaller profiles simply using published accounts".

of Globalisation on National Accounts (UNECE, 2011). OECD globalisation indicators were produced in 2005 and 2010 while UNECE produced a Guide to Measuring Global Production in 2015 (UNECE, 2015a).

However, contributions have been generally referred to single statistical domains and mainly devoted to measure *ex post* the impact of MNEs operations according to a national perspective. For example, the profiling technique has been mainly focussed on establishing the appropriate statistical units (Eurostat Business Register Recommendation Manual, BRRM 2010) rather than assessing economic variables measurement. Therefore, taking advantage of the experience of some pioneering countries, such as Ireland, the Netherlands, Sweden, Finland (UNECE, 2011), and Canada (Statistics Canada, 2018), the idea of setting up in the organisational chart such specific units inside dedicated to the largest and complex MNEs, alongside other activities, has started to be considered also in different contexts.

Aiming at identifying the effects of MNEs' complex restructuring at the early stages of the statistical production process, the Istat Modernisation Programme in 2016 established within the Directorate for Economic Statistics (DCSE)<sup>5</sup> a Large Cases Unit (LCU) Team responsible to deal with the inconsistencies of the largest MNEs operating in Italy. Therefore, the main ambition of the team is to keep the statistical domains experts informed by adopting an *ex ante* approach applied to a National Statistical Organisation (NSO) level.

The main objective of this paper is describing the results of several analyses the LCU team has conducted during the first phase of the project in evaluating size, significance and complexity aspects (UNECE, 2015a) according to both the characteristics of the national economy and the internal organisation of Istat. Size aspect defined according to one or more variables, such as gross value added, turnover, value of production or number of employees, helps in identifying largest MNEs. Since larger does not necessarily mean complex, the complexity aspect needs to be evaluated according to other features such as the enterprise group structure or the number of activities carried out by the MNE. Finally, significance for data collection is related to the selection of the most important respondents, the dominant groups or the respondents who

<sup>5</sup> Official name in Italian "Coordinamento delle unità economiche complesse" (Coordination of complex economic units).

have problems in data provision. Therefore, the work is fully embedded in the context of official statistics studies and, in particular, it shows the process and rationale of the Italian experience in setting up an LCU team.

Several analyses were conducted investigating two basic concepts: the contribution to the National statistics of the MNEs in the target population (large and complex MNEs) that combines size and significance aspects, and complexity, proxied by the multi-business concept and business groups restructuring signals. In addition, conceptual external inputs from Eurostat and other international organisations were considered to improve the international perspective.

In terms of contributions, at first the paper had a practical implication: findings helped in defining the daily activities to be carried out by the Italian LCU team. However, findings also added new insights and suggestions to the international debate on LCUs or similar activities sharing our national experience in the first phase of the project.

The paper is organised as follows. Section Two is intended to review the most significant literature, the international manuals and reports on different statistical domains and the countries best practices related to MNEs and LCU or similar activities. Basic concepts under investigation during the early stage of the project are discussed in Section Three, while Section Four presents the preliminary studies results. Interesting international initiatives for our purposes are shown in Section Five. Conclusions, limits and possible future research developments are presented in Section Six.

## 2. Background and context

UNECE, 2011 guide represents the main output of the Expert Group on the Impact of Globalization on National Accounts established in 2007 to review and present proposals for dealing with the main problems in the compilation of national accounts and related source statistics due to globalisation. For the purposes of this paper, the guide contains in nuce two important elements that became crucial in the following years. First of all, the latter reference to the 'related source statistics' clarifies that the treatment of globalisation related issues posed by the MNEs is a process that has to start at the beginning the business register and related statistics. A coordinated approach is almost essential and efficient in handling large and complex MNEs. Secondly, the Guide illustrates the experience of four countries, Ireland, the Netherlands, Finland and Sweden, in dealing with measurement issues and statistical units definition problems (van Delden et al., 2018) associated with the global nature of the MNEs. In these cases, NSOs have adopted an organisational solution by establishing dedicated units, named at that time as 'consistency units' and which will be later generally identified as LCUs, regarded as "essential for ensuring that the data of large corporations are incorporated coherently across economic statistics" (UNECE, 2015a: 87).

Generally, these teams/units are composed by experts with different skills adopting a micro-data approach. As part of the overall objective of addressing globalisation problems caused by MNEs, the approaches contain some useful differences to explain the multi-faceted issue of concerning coherence. Countries' specific aims, such as ensuring the coherence and plausibility in compiling national accounts and related statistics (Ireland), identifying inconsistencies at an early stage of statistical production to resolve them and to prevent their recurrence (the Netherlands), asking the right questions to the MNEs (Sweden), comparing monthly, quarterly and annual data within and across areas of statistics (Finland), helped us in defining our approach.

The need for this organisational solution firstly emerged in countries very dependent on international trade and investments or in which very few business groups cover a large proportion of the national economy. According to UNECE, 2011, top ten foreign-owned MNEs in Ireland covered 34 per cent of all exports of goods and service, while top 50 companies in Finland account for about 30 per cent of turnover. Furthermore, top 320 enterprise groups in

the Netherlands cover 40 per cent of turnover and 35 per cent of gross value added (UNECE, 2015a). Other interesting examples of LCU activities outside Europe include the Enterprise Portfolio Management (EPM) programme, responsible for managing all aspects of ongoing relations with Canada's 300 largest and most complex businesses that started in 2006. In their experience, 99 per cent of the entities recorded in their Statistical Business Register are referred to as 'simple' (one operating entity, one industry, and located in one geographic region), however the remaining one per cent of businesses labelled as 'complex' account for approximately 52 per cent of the total economic activity in Canada (Statistics Canada, 2018).

Concerning organisational aspects, Chapter 6 of the Guide to Measuring Global Production (UNECE, 2015a) focussed on presenting the outcomes of a survey carried out in March 2013 by the Task Force on Global Production to get a better understanding of the organisation of work and the type of analysis carried out by LCUs. The results showed as criteria for identifying large and complex entities generally comprise the following aspects: size, complexity and significance, in terms of selecting the most important respondents. Furthermore, concerning the organisation of the activities, despite NSOs have generally set up independent units, there are also teams of experts (Finland) and mixed forms (New Zealand). Such kind of units have been variously located: business statistics (Hungary); business statistics that also include national accounts (Canada, France and the Netherlands)); national accounts (Ireland), and data collection (Finland and Sweden) (UNECE, 2015a). Finally, LCU attributions vary greatly among different experiences including activities such as monitoring to identify current and forthcoming problems in data provision, data analysis, profiling, coordination, and contacts with large respondents. Despite the location of LCU (or similar) being recommended to be placed near the start of the process, maybe sitting alongside the statistical business register, the most important aspect to consider is the variety of skill set required to undertake the work.

However, the acceleration of the international debate on LCUs and similar type of activities occurred following the so-called 'Irish case'. In July 2016, the Irish NSO communicated for National Accounts for 2015 an annual increase in GDP of 26.3 per cent (CSO, press release, 2016) reporting a scale

of increase that was unprecedented in OECD economies historically<sup>6</sup>. As commented in the press release, in that case, globalisation exposed all its pervasive effects on official statistics showing as Gross Domestic Product (GDP) and Gross National Income (GNI, previously GNP) do not always help to understand what is happening in a single economy. As stated by Stapel-Weber and Verrinder (2016), "This is the first time — despite long conceptual discussions — that statisticians have been prompted by real economic events to take globalisation as seriously as it has to be taken, with major impacts for all kinds of economic statistics (national accounts, balance of payments, business, employment and trade). What makes the case very relevant for Europe, and the work of Eurostat, is that this could happen again any time as huge multinationals move their business around Europe and the globe".

In the light of these premises, a dedicated team devoted to large and complex MNEs operating in Italy was introduced in Istat during 2016 according to the reorganisation programme. Identifying the inconsistencies or changes required at the earliest possible stage of the statistical production process and sharing all the issues related to the MNEs investigated have been identified as the main objectives of the LCU Team. The activities of the team, partly ascribable to the LCU typology, has been included in the official mandate of the Directorate for Economic Statistics. Co-operating and collaborating with other Directorates, such as National Accounts to obtain a common and coherent treatment for specific and complex cases, and Data Collection to evaluate the degree of cooperation of the main units, have also been considered as key elements to the process.

According to the internal organisational guidelines, members of the LCU Team have been engaged part-time in the work and selected on the basis of different experiences in order to ensure both an appropriate mix of skills and an extensive network across different divisions. Heads of business primary statistics divisions have been consulted to identify staff to be involved. In particular, the team has been made up of experts of National Statistical Business Registers (nSBR), Structural Business Statistics (SBS), Short Term Statistics (STS), Foreign Affiliates Statistics (FATS) and Foreign Trade statistics. Therefore, experts from all the statistical domains within the Directorate

<sup>6</sup> Central Statistics Office. https://www.cso.ie/en/media/csoie/newsevents/documents/IrelandEconomicGrowthFigures.pdf

for Economic Statistics have been included in the team. Concerning the resources, team members have been involved from a minimum of five per cent to a maximum of 70 per cent of their working time and, during these years, the average number of different experts in the network has been 15 while the maximum has been 23. Finally, no additional budget and funds have been used.

### 3. Methods

During the first phase of the project, the LCU Team conducted several studies for evaluating what large and complex could mean in our context. According to Task Force on Global Production survey results (UNECE, 2015a; see previous Section), population of large and complex enterprises are generally identified by the following aspects: size, complexity and significance. The size of a MNE could be measured by several variables, such as gross value added, turnover, value of production and number of employees. Complexity could be assessed by the number of legal units (or enterprises) in the group or the number of layers in the structure of the enterprise group. Furthermore, complexity could be evaluated in terms of operations by industry or geographical area. Finally, significance for data collection could be achieved by selecting the most important respondents, the dominant groups or the respondents who have problems in data provision.

Based on these three criteria – size, complexity and significance – the LCU Team defined the number of top MNEs in the target population (from hereon will be referred to as 'target'), that is the list of the largest and complex MNEs acting in Italy, and investigated two basic concepts:

- the contribution of top MNEs to the National statistics, that combines size, useful to define the target, and significance, an important aspect in understanding the impact of groups under analysis on business surveys.
- ii) the complexity proxied by the multi-business concept and by restructuring signals from business groups.

Figure 3.1 shows all items evaluated by the team for each topic.



#### Figure 3.1 - Basic concepts and methods

Source: Authors' elaboration

Concerning the first item, nSBR<sup>7</sup> and National Statistical Business Register on Enterprise Groups<sup>8</sup> were examined to develop a provisional procedure to select top business groups operating in Italy to be included in the target list. Therefore, the size topic was investigated through the evaluation of the impact on statistical business registers. Furthermore, an analysis of the actual role of large business groups from a register-based perspective was made (Paragraph 4.1). In addition, the LCU Team conducted studies to verify the contribution and the degree of cooperation of top business groups on short-term surveys. These analyses helped in defining the actual role of large business groups in the composition of two short-term economic indicators, the monthly retail trade index (Paragraph 4.2), and the services turnover index (Paragraph 4.3).

Regarding complexity, we chose to emphasise economic aspects and breaking events rather than other elements, such as the number of layers in the structure of the enterprise group and complexity in terms of geographical area. Nevertheless, those latter aspects are currently captured by other Istat teams devoted to profiling and EuroGroup (EGR) activities.

The LCU Team firstly focussed on the concept of multi-business activities. In order to evaluate the presence of the kind of activity units (KAUs) for top enterprises (Paragraph 4.4), a qualitative study was conducted to analyse signals deriving from the SBS Total Survey on large enterprises (SCI)<sup>9</sup> and financial statements (administrative data). According to the Council Regulation (EEC) No 696/93 on the statistical units, a KAU groups all the parts of an enterprise contributing to the performance of an activity at class level (four digits) of NACE classification and corresponds to one or more operational subdivisions of the enterprise. Furthermore, within the framework of 2011 Business Census, a dedicated survey on top Italian enterprises (or enterprise groups)<sup>10</sup> (MPSUC) was launched to collect information on their organisation and the LCU Team investigated the results concerning the number of different business lines carried out (Paragraph 4.5). The concept of business line considered for this survey derives from the international accounting standard IRFS 8 concerning the disclosure of the operating segments in the financial statements. An operating segment is defined as a component of an

<sup>7 &</sup>lt;u>https://www.istat.it/it/archivio/216767</u>

<sup>8</sup> https://www.istat.it/it/archivio/234313

<sup>9</sup> https://www.istat.it/it/archivio/13635

<sup>10</sup> https://www.istat.it/en/archivio/187836

entity (enterprise or enterprise group) that "engages in business activities from which it may earn revenues and incur expenses (including revenues and expenses relating to transactions with other components of the same entity); whose operating results are reviewed regularly by the entity's chief operating decision maker; for which discrete financial information is available" (IFRS 8:2).

Restructuring events inside the target groups were considered as another way to proxy the complexity inside MNEs (Paragraph 4.6). The LCU Team defined different streams of monitoring activities to share signals concerning complex reorganisations able to affect the statistical production.

Conceptual external inputs from Eurostat and other international organisations were considered to refine the definition of the tasks also according to an international perspective. Cross-countries impact of MNEs activities requires the development of international networks able to guarantee information and coordination, such as the Eurostat Early Warning Signal network (Paragraph 5.1). As international aspects tend to be very demanding, The LCU Team has started following the news concerning these topics trying to define dedicated tasks to handle them (Paragraph 5.2).

## 4. Findings

This section includes an overview of the analyses produced by the LCU Team during the first phase of its activity. These works were designed to define the target MNEs, investigating their role and evaluating their complexity.

## 4.1 The target groups and their impact on nSBR

The size topic was investigated through the evaluation of the impact on statistical business registers. The LCU target list was defined on the basis of the Italian Business Register and the Italian Business Register of Enterprise Groups data. The procedure followed for a year 't' starts at the business group level. Two different orders for business groups are generated, by turnover and number of employees. After that, groups belonging to the top 200 ranking both for turnover and number of employees are selected. The third step consists in completing the list to obtain a total number of 130 business groups according to the turnover order only. Finally, worldwide top MNEs operating in the socalled digital economy, such as e-commerce and sharing economy, are added in the list even if their values are not very high in Italy. The total number of business groups annually included in the target is 140, 45 per cent foreignowned MNEs, 41 per cent of Italian MNEs, and 14 per cent of domestic groups. Italian MNEs cover the 58 per cent of total turnover and 53 per cent of employment, while foreign-owned MNEs generate 35 per cent of turnover and 27 per cent of employment. Domestic groups account for 20 per cent of employment and seven per cent of turnover.

Enterprise groups in the target are formed annually and in average by 3,300-3,600 legal units<sup>11</sup>. By comparing years 't' and 't-1', generally, about 75 per cent of the enterprises appear in both the targets while half of the remaining units are not confirmed because of inactivity or liquidated in the year 't'. A longitudinal evaluation is also carried out to test the stability of the business groups in the target.

Due to the characteristics of the Italian system, mainly based on small and medium enterprises, statistical registers annually record 4.4 million

<sup>11</sup> For the reference year 2016, according to the nSBR outputs, legal units coincide with the statistical unit "enterprise".

enterprises and about 220,000 of those belong to business groups (nearly 100,000). Therefore, coverage percentages found for other Countries as shown in Section Two (UNECE, 2011; UNECE, 2015a; Statistics Canada, 2018) are unexpected in our case. Enterprises in the target list represent annually less than one per cent of the total units in the nSBR. In terms of economic variables, they cover the eight per cent of total employment, 16 per cent of gross value added, and 26 per cent of total turnover. Considerable differences have emerged among NACE Sections and adopted criteria. Table 4.1 shows the NACE Sections greatly covered by the enterprises included in the LCU Team's target list.

Finally, it is important to consider that Section K (Financial and insurance activities) has been evaluated over the employment criterion only, showing a coverage value of about 40 per cent, while the contribution of top MNEs for Manufacturing (Section C) accounts for about eight per cent of total employment, nearly 18 per cent of turnover, and more than 11 per cent of gross value added. Employment criteria generally shows a great degree of stability over time, while turnover and gross value added are less stable.

 Table 4.1 - Enterprises in the LCU target list: criteria for the evaluation of the impact in terms of economic variables. Main NACE Sections. Year 2016

NACE Sections	Employment	Turnover	Value Added
B Mining and quarrying	46,0	92,2	58,6
D Electricity, gas, steam and air conditioning supply	63,5	76,1	68,6
E Water supply; sewerage, waste management and remediation activities	17,8	22,4	26,5
H Transportation and storage	27,3	25,7	39,7
J Information and communication	19,1	45,7	41,8
H Transportation and storage J Information and communication	27,3 19,1	25,7 45,7	

Source: Authors' elaboration on Istat data

### 4.2 Contribution of top groups to STS monthly retail trade index

The analysis focusses on the contribution and the degree of cooperation of large business groups to the Monthly Retail Trade Survey and specifically assesses the weight of large business groups falling into LCU target list on total turnover in the retail industry. For this study, in order to determine the weight of the units belonging to the LCU target list, SBS data on turnover were preferred against information from the survey, as values gathered through questionnaires included goods with different VAT rates and missing data. Furthermore, using absolute values from the survey is not recommended as indicators of retail trade rely on algorithms calculating percentage changes; therefore, to obtain absolute values, the whole information system of the survey should have been recalibrated significantly.

Restructuring of groups and enterprises could determine missing data for new units to be potentially included in the sample. When such an issue occurred, missing values were imputed using administrative sources (financial reporting statements) and SBR data.

Considering that the study was conducted during the initial stage of the project (end of 2016-beginning of 2017), in order to use structural information on turnover as previously stated, the selected reference year for analysing data was 2014. Businesses involved in the LCU target were merged to the sample of the Monthly Retail Trade Survey that for the reference year counted about 8,000 enterprises resident in Italy, classified as operating in NACE Division 47.

Results highlight that about 30 per cent of the business groups in the LCU target list were sampled in the Retail Trade Survey, having at least one enterprise for each group. The enterprises belonging to the target business groups were 89, representing just one per cent of the sample, however they contributed for 50.4 per cent of the total turnover of the retail trade sample for the year 2014. Another 23 enterprises belonging to LCU target list were not selected for the sample even though they were eligible, however their potential contribution to total turnover accounted approximately for a further two per cent.

Generally speaking, large groups demonstrated a very active cooperation and showed a high response rate, especially larger enterprises in terms of employees. The analysis revealed that eight groups (out of 40) partially cooperated, as several enterprises answered the survey, while other enterprises from the same group did not provide the required data; just one of the nonrespondents was a very large enterprise. Concerning this latter case and for two more enterprises, the top management of the business groups officially declared to Istat that their policy does not allow them to share their sales figures (even when surveys are mandatory). Failure to complete and return a compulsory questionnaire results in legal action handled by a designated Istat division. An enterprise group refusing to cooperate represents a crucial (and actually the most negative) signal to be considered by the LCU to decide whether to build relations with the businesses.

### 4.3 Contribution of top groups to STS turnover services index

This work has the objective to verify the impact and the degree of cooperation of the business groups in the LCU target list for service industries in complementing the analyses based on size dimension (Paragraph 4.1). Enterprises in the sample of the quarterly survey on Turnover Services (FAS) for 2014 were approximately 21,500, of which about 7,100 belonging to business groups for the same year. Concerning the LCU target list, 393 enterprises were sampled in the FAS survey (about 11 per cent of the enterprises in the target list) belonging to 98 business groups (the 70 per cent of the groups in the scope).

The degree of cooperation was evaluated considering the quarters of the year 2014. A four-digit flag was introduced according to answers enterprises gave for each quarter (responses and non-responses were coded in all possible combinations of '1' and '0'): 360 enterprises (92 per cent of the target) answered for all the quarters; 12 for one to three quarters (3 per cent); 21 did not provide data for any of the four quarters (5 per cent). The involvement of top groups operating in Italy was evaluated in terms of number of enterprises in the sample. For nine groups, more than ten enterprises belonging to the same group were investigated, reaching the maximum of 24 in one case. For 18 cases, from five to ten enterprises of the same groups were sampled and another 46 groups were surveyed for more than two (and less than five) enterprises. Finally, 25 groups were represented in the sample by one enterprise only.

Considering the enterprise group level, the study highlighted an excellent degree of cooperation in the FAS survey. For the largest 20 groups, more than 90 per cent of their enterprises cooperated for all the quarters. Moreover, 48 groups (49 per cent of the business groups involved) investigated through

more than two enterprises answered for all the enterprises and all the quarters. In addition, it is also important to underline that the majority of the cases of missing answers for one or more quarters were generally justified by demographic events (mergers, split-offs) mainly concerning enterprises within the same groups and already in the sample. Only in two cases, a complete non-response was registered.

In terms of contribution, the enterprises belonging to the LCU target list that answered for all the quarters of the year 2014 represented 33.5 per cent of the total turnover. Also, it is important to note that for the service industries, the group dimension plays a crucial role. In fact, the respondents to the survey for all the quarters belonging to business groups covered for 2014 about the 88 per cent of the total turnover recorded for the survey.

# 4.4 Financial reporting and KAUs

This section summarises the results of the analysis of a small number of large companies belonging to enterprise groups in the LCU target list in 2016. The aim is to investigate the presence of information about different economic activities coexisting within the same company to identify potential KAUs.

Data from the SBS Total Survey on large enterprises (SCI) referred to the year 2015 were analysed in order to evaluate the presence KAUs and the results indicated that less than one per cent of the enterprises (86) provided economic data on this statistical unit. SCI is annual survey concerning the economic accounts of the enterprises, including professional and artistic activities, with 250 or more persons employed.

In order to identify complex business organisations, the 30 largest companies in terms of both gross value added and turnover were considered and then, eight of those which did not declare any KAUs in the SCI survey were selected for further analyses.

A more in-depth study of such companies was carried out by consulting different administrative and statistical sources. Financial statements from the business accounts represented the main source of information, especially sections related to Income Statement, notes to the financial statements and Management Report. The nSBR was also examined to check whether information found in administrative sources had already been acquired by Istat.

Among the selected companies, one performs its principal economic activity in wholesale and retail trade industries, four in transportation and storage industry, two in manufacturing industry and one in the information and communication industry. For these selected companies, data from financial statements led to identify 19 potential secondary economic activities: 12 of them were both in the administrative and statistical sources, the remaining seven were not in the nSBR. Even if referable to a small sample, it is interesting to note that an average number of 2.4 activities were disclosed in the financial statements.

It is recognised that the use of data from business accounts in economic statistics can be challenging and often requires adjustments to move on to the basis required for business statistics and national accounts. Even though the analysis provided useful signals of different business lines coexisting in complex enterprises, the disclosure of this kind of information in the financial statements is strictly dependent on the reporting choices adopted by each enterprise. Furthermore, the comprehension of such information was not always immediate. In some cases, financial statements contain separate income statements for different business lines and identifying the secondary economic activities was easy. Details on revenues and operating costs could also allow to easily determine the economic importance of each activity and its contribution to the overall business income. In other cases, financial statements contain the overall income, and the existence of secondary economic activities could be only presumed from the Notes to the Financial Statements or from the Management Report.

In addition, this analysis pointed out that the statistical information gap on operative and organisational aspects characterising companies that are particularly relevant on the Italian scenario. Seven potential secondary economic activities identified through administrative sources were recorded neither in the Survey nor in the nSBR and three of them were closely related to the company's mission and economically significant. The remaining four potential secondary economic activities do not characterise ordinary operations and marginally contribute to the overall revenue of the single enterprise but, in some cases, they could have a significant impact in terms of industries involved.

## 4.5 Business Lines and complex units (MPSUC survey)

Within the framework of 2011 Business Census, in 2014, Istat conducted a multi-purpose survey on top Italian enterprises or enterprise groups (MPSUC). The objective of the survey was measuring the characteristics of the resident companies that present a considerable degree of complexity in organisational models and decision-making processes.

The reference year of the MPSUC survey is 2013 and about 3,000 complete, coherent and useful questionnaires for publication were collected (response rate of 90.2 per cent).

Enterprises and business groups were split into the following three subgroups:

- companies or main groups (A): companies or groups of companies that in the three-years period 2011-2013 achieved, on average, a turnover equal to or greater than 2.5 billion euros or that had at least 10,000 employees.
- companies or large groups (B): companies or groups of companies that in the three-years period 2011-2013 achieved, on average, a turnover lower than the values recorded for the main companies or groups but in any case, equal to or greater than 250 million euro and which had at least 500 employees (but less than 10,000).
- medium-large companies or groups (C): other companies or groups other than the main and large companies that are included in the survey sample according to the defined criteria.

For the respondents, 4,680 business lines able to show the complexity aspect (see Section Three) were collected. For what concerned this survey, a business line could be defined according to the following criteria: identifying a relevant and identifiable segment of the economic activities carried out within the company; generating revenues and costs; information available in a structured and non-occasional way on the costs and revenues generated by the individual business line; operating results periodically reviewed at the highest operational decision-making level. According to the results, respondents generated 1.6 business lines on average. The complexity of the companies increases with the size: the companies or main groups presented on average about three business lines in Italy and one business line abroad; the companies or large group presented on average one business line in Italy and one business line in Italy and one business line abroad.

It is important to underline that 71 per cent of the respondents declared to operate through one business line. The rest, about 850 enterprises/groups, carried out their activities running more than one business line (in total, about 2,500). Thus, these respondents operated with an average number of three business lines. For the sub-groups, the situation was as follows: for group A, the average number was 3.5; for group B, 3.4; and for group C, 2.8.

Concerning the criteria used by the groups/enterprises to define the business lines, table 4.2 synthesises the answers for the three sub-groups. Different types of products/services (62.5 per cent) and different production processes (27.6 per cent) represented the main criteria for defining the business lines covering more than 90 per cent of the cases. Different market location mainly affected main groups while different customers played an important role especially for groups A and B.

SUB-GROUPS		MAIN CRITERION							
	Different production processes of pro- ducts/services	Different types of products/services	Different mar- ket location	Different customers	Others	Total			
Group A	24	58,7	9,3	8	-	100			
Group B	27,1	57,8	4,7	9,7	0,8	100			
Group C	28,3	64,8	2,4	3,4	1,1	100			
Total	27,6	62,5	3,6	5,3	0,9	100			

 Table 4.2 - Criteria for the definition of the business lines by economic dimension (percentage values). Year 2013

Source: Istat, Multi-purpose survey on top Italian enterprises or enterprise groups (MPSUC)

Furthermore, in order to assess the degree of stability of the business lines reported in the MPSUC survey, 16 of the most relevant business groups within subset A were selected and further scrutinised for the period 2013 - 2016.

Consolidated financial statements were the main sources used to compare the results of the MPSUC survey. Statistical units were linked to the Italian Business Register of Enterprise Groups in order to evaluate the hierarchical control chain position of the respondent within the enterprise group structure. When applicable, a comparison was made between business lines collected via survey and operating segments (IFRS 8) disclosed in the consolidated financial statements.

During the period 2013 - 2016, in seven cases (out of 16) the number of business lines did not change while in four cases changed due to internal reorganisations (two increases and two decreases). A very complex organisational change was found in one case because the group adopted a Countries/Divisions matrix in comparison to the previous one based on business lines. Finally, in four cases, differences were found also for the reference year 2013 even if not very significant in terms of number of employees and turnover. In these cases, the number of business lines indicated in the MPSUC survey differs from those indicated in the consolidated financial statements for two reasons: for the purposes of the survey, respondents merged business lines disclosed separately in the financial statements; holding activities was not considered as business lines in the survey.

Results show that more than a half of the business groups in the sample modified their structure in terms of disclosure of business lines during a period of three years.

# 4.6 Reorganisation signals

To investigate complexity, the LCU Team decided to mainly focus on the analysis of restructuring signals. The basic idea is to analyse signals in a wider perspective considering business groups or sub-sets of them.

Firstly, an evaluation could be carried out by analysing the requests collected through the Business Statistical Portal which is the infrastructural solutions Istat has adopted as single access point to data acquisition systems. The web device also allows respondents to send their requests to the survey and registers managers.

Concerning our purposes, analysing this kind of requests has a crucial advantage: data are stored in a dedicated repository and can be immediately processed. In fact, the records contain internal identification codes that link them to the nSBR. Therefore, evaluating the requests for the enterprises belonging to the business groups in the target offered the chance to build a provisional database to retrieve indications on potential complex restructuring of the main groups active in Italy. In that way, the analysis of the signals determines an ex post evaluation of the cases.

Secondly, according to the experience of the colleagues working on FATS statistics, a further possibility to early discover complex restructuring events resides in analysing articles in specialised newspapers and magazines and public sources. The aim is to capture relevant signals about events that may significantly change the structure of the main enterprises of business groups with large implications at national and international level. The main source for this activity is '*Il Sole 24 Ore*' which is the main Italian daily business newspaper, but also other specialised magazines could be considered. In this case, the activity could be assimilated to an *ex ante* evaluation of the signals.

Due to this latter characteristic, the activity is fully consistent with the main purpose of the project that is identifying the effects of complex restructuring of MNEs at the early stages of the statistical production process. At the same time, the main risk consists in sharing news of a poor relevance for statistics or not completely defined. To reduce this risk, the activity should consist in evaluating and then sharing public news when they are certain, with significant effects from an identifiable starting point, and able to influence the statistical outputs or affect the contact activities with the large and complex units. In addition, another important risk is related to the fact that news must be firstly codified to identify the units affected by the event.

This study resulted in defining the process that the LCU Team should follow to monitor day-to-day activities starting from publically available sources (figure 4.1).



Figure 4.1 - Monitoring from publically available sources - the process

Source: Authors' elaboration
#### 5. International aspects

This section includes the description of external inputs deriving from international projects and initiatives that helped LCU team defining part of their day-to-day tasks.

#### 5.1 Eurostat Early Warning System (EWS) network

Globalisation has created new business opportunities for MNEs and challenges for NSOs in producing effective business statistics and National Accounts aggregates. For instance, a rapid record of a business reorganisation of an MNE has become particularly difficult in terms of data collection, production, and statistical analysis, especially when crossing national boundaries and having sizeable effects on the European (and beyond) and National Statistical system. In response to the needs of NSOs, in 2017, Eurostat established the Early Warning System<sup>12</sup>, to ensure broad consistency in applied methods, statistical treatment and communication of results across the European Statistical System (ESS). The EWS has been conceived as a structured, light developing procedure based on the voluntary cooperation among National Authorities dealing with official statistics. Coordinated by the Eurostat EWS-secretariat, the network of national EWS correspondents aims to early exchange information on restructuring events across Member States directly concerned, by achieving an agreed methodological treatment of the cases. As National Accounts main aggregates and the Principal European Economic Indicators (PEEIs) are concerned, NSOs are Members of the EWS. Moreover since 2018 Eurostat has been cooperating with the European System of Central Banks (ESCB) and the European Central Bank (ECB), which has become part of the EWS network of correspondents.

The timely and interlinked communication among EWS Members, respecting the statistical confidentiality, ensures that the restructuring cases are treated consistently addressing issues related to business statistics and National Accounts revisions. Hence, the individual cases are discussed by an *ad hoc* task force composed by national correspondents of the Countries

<sup>12</sup> Eurostat, 'An Early-warning System (EWS) for the correct and consistent statistical treatment of restructuring events of multinational enterprise groups and their enterprises in European statistics', March 2020. https://ec.europa.eu/eurostat/web/economic-globalisation/early-warning-system

affected by the globalisation event. Moreover, the EWS correspondents<sup>13</sup> of the Countries and Eurostat share only necessary and limited information on the specific restructuring case when dealing with consistent statistical treatment of single globalisation cases. NSO reports containing the description of the restructuring case, information on affected statistical domains, the list of other Countries which may be concerned and methodological assessment are stored by Eurostat in the secure collaborative platform used for sharing information and resources (S-Circa BC). The general access to this group is restricted to EWS correspondents and the case-specific information are available only for the NSOs that are directly involved. As a result of the case discussions, Eurostat produces an anonymised methodological summary note containing the case statistical treatment agreed by EWS Members concerned.

To develop further methodological and conceptual guidance on restructuring of MNEs, Eurostat is planning to involve the OECD in the EWS, to address globalisation events adding further value, when non-EU countries are affected.

At national level, the Italian Institute of Statistics is committed to EWS through the LCU Team. To identify and to analyse the EWS cases related to MNE restructuring and globalisation issues, Istat has set up a small group of experts in various domains within the LCU Team. Within the framework of EWS, the LCU team works closely on business statistics by liaising with National Accounts experts, to ensure that EWS correspondents appropriately deal with MNE cases. By capturing the impact of globalisation through business statistics and national accounts aggregates, the multi-skilled team are involved by EWS correspondents when the MNE case needs to be analysed or detected.

## 5.2 Other international initiatives

As described in Section 2, the wider-international dimension of the MNEs requires an answer that goes beyond the national perspective (or EU perspective) not only to analyse the restructuring events, but also to deepen the knowledge of the global operations of the largest business groups, regardless of organisational changes.

<sup>13</sup> The EWS network consists of two national correspondents nominated by each National Statistical Institute and each International Authority dealing with statistics.

Concerning the former aspect, the main answer of the ESS has been the development of the EWS (Paragraph 5.1) while, for the latter, Eurostat has launched several initiatives to analyse the current treatment of MNEs among different countries. Studies focussed on implications of MNEs activities in compiling national accounts and the GNI MNE Pilot exercise, launched by the Joint BSDG/DMES14 Task Force on globalisation, falls specifically in this stream of analyses. Considering the purposes of the GNI MNE Pilot exercise, it is interesting to note that the work of the LCU requires a coordinated and consistent approach to different tasks, such as EGR, EU Profiling, and all the other initiatives related to LCU programmes (Alajääskö et al., 2018). Furthermore, mainly for National Accounts purposes, other actions, such as the Integrated Global Accounts and Global Production (IGA) and the Full International and Global Accounts for Research in Input-Output (FIGARO) projects, dealt with the globalisation-related aspects and their possible implications to the accounting frameworks at EU and national levels. Regarding large and complex enterprises, the Voorburg Group on Service Statistics<sup>15</sup> also dedicated a specific poster to the topic in the meeting hosted in Rome in 2018<sup>16</sup> while, in the same year, the Working Party on International Trade in Goods and Trade in Services Statistics emphasised the increasing use of LCUs solution within the NSOs to better understand complex issues<sup>17</sup> such as globalisation and digitalisation.

The Global enterprise perspective introduced in the UN volume "The Guidelines on Accounting for Global Value Chains: GVC Satellite Accounts and Integrated Business Statistics" (2009) has also shed new lights on the way a MNE can be analysed in a wider perspective. In fact, according to the business statistics framework introduced by these guidelines, the global enterprise activities can be broken down first by business line and then by the supporting business functions, which together define the business process. In addition, each business function of a business line can be carried out inside or outside the global enterprise and can be located either in the resident country or abroad (UN, 2019 Part three, Integrated business statistics). Such a framework could be very useful in complementing the complexity analyses in

<sup>14</sup> BSDG: Business Statistics Directors Group; DMES: Director of Macro-Economic Statistics.

<sup>15</sup> https://www.voorburggroup.org/

<sup>16</sup> https://www.istat.it/en/voorburg-2018

<sup>17</sup> http://www.oecd.org/officialdocuments/publicdisplaydocumentpdf/?cote=STD/CSSP/WPTGS(2018)3& docLanguage=En

defining the groups in the NSO target. However, different linkages are needed between the business functions and the industrial / product classifications used within NSOs.

Finally, different proposals have been made for the establishment of an LCU network. The work of the UNECE "Task Force on Exchange and Sharing of Economic Data", established by the Bureau of the Conference of European Statisticians (CES) in 2017, highlighted the growing need to set up and coordinate an international network of experts on MNEs in order to exchange experience and best practices in data sharing, and metadata type information on MNEs, such as their structure and the available sources. The network could also improve the sharing of information on tools and techniques.

Eurostat has investigating the topic of an LCU network launching new projects also considering different forums of related aspects such as profiling and macroeconomic statistics. The benefits of such a structure could be summarised as follows: "An international network of LCUs seems to be the right answer to facilitate the necessary exchange of relevant information. LCUs may have a special role in facilitating international work towards better understanding MNE groups. LCUs could provide a point of contact between NSOs for MNE group profiling and data exchange. The active involvement of LCUs in international work and the follow-up of results achieved in recent international initiatives would support national work on MNE groups" (Hussain *et al.*, 2019: 54). Because of the importance of the ongoing international initiatives, the LCU team members have been variously involved in different projects in order to follow the advancements in these fields.

## 6. Discussion and conclusions

During the first period of its activity, Istat Large Cases Unit Team conducted several studies in order to determine the business groups in the target list, their contribution to the National statistics, combining size and significance aspects and their complexity, proxied by the multi-business concept and business groups restructuring signals. International projects were also considered to follow the international debate. Therefore, the organisational solution Istat adopted to deal with the inconsistencies of the top MNEs involved setting up a dedicated team and improving the connection to global networks, such as the Early Warning System (EWS).

Analyses were conducted to examine both the characteristics of the national economy and the internal organisation of Istat, focussing on the team mandate that relies on identifying the effects of complex restructuring of MNEs at the early stages of the statistical production process.

Business groups in the target were defined in terms of size, based on turnover and employees, and their impact on national statistics. The result of the selection method consisted in 140 business groups in the target per year. Due to the characteristics of the Italian system, mainly based on small and medium enterprises, the enterprises in the target represent annually less than one per cent of the total units in the nSBR, the 8 per cent of the total employment, the 16 per cent of the gross value added, and the 26 per cent of the total turnover. Business groups have a greater impact on the two STS surveys considered for this study, contributing for more than a half of the total turnover for the Monthly retail trade survey and more than one third for the quarterly survey on Turnover Services. These studies also show an excellent degree of cooperation of the top business groups in the target in answering the NSO's surveys.

Considering the results of these studies, the characteristics of the LCU based on a network of experts working part-time at the project and the presence in the organisational chart of other initiatives related to globalisation aspects, the number of 140 business groups for monitoring activities appears to be feasible and also confirmed for the following years.

Complexity was firstly investigated by the multi-business concept. An analysis combining statistical and administrative sources was conducted and even if associated to a small sample of enterprises, an average number of 2.4 activities were disclosed from financial statements. In total, 19 potential secondary economic activities were found: 12 of them were both in the administrative and statistical sources while the remaining seven were not in the nSBR. To consider that none of them reported KAUs data in answering the SBS Total Survey on large enterprises (SCI). Concerning the same topic, according to the multi-purpose survey on top Italian enterprises or enterprise groups (MPSUC), 71 per cent of the respondents declared to operate with one business line but about 850 enterprises/groups carried out their activities with more than one business line (an average number of three business lines). It is very interesting to note that the number of business lines increases by the size of the units: 3.5 for the main groups; 3.4 for large groups; 2.8 for medium-size groups. These studies show that the presence of several activities within the same companies is a very frequent feature and that it is even more relevant for larger companies. Different types of products/services (62.5 per cent) and different production processes (27.6 per cent) represent the main criteria for defining the business lines covering more than 90 per cent of the cases. However, the complete evaluation of the different businesses carried out by an enterprise is a demanding and time-consuming activity requiring an indepth study of the financial reports that generally are available at least six months after the reference period. For this reason, in order to investigate the complexity at an early stage of the process, the LCU Team focussed on analysing restructuring signals in a wider perspective, by considering business groups. According to these findings, LCU monitoring activities should be based on processing different signals to obtain a complete comprehension of the business reorganisation and to share a quick assessment of the impact for the statistical domains. Requests collected through the Business Statistical Portal, news from specialised newspapers and magazines and Eurostat EWS cases, emerged as the main research streams useful to consider in defining and carrying out the LCU monitoring activities.

The work underlying the paper – setting up a specific unit inside the organisational chart dedicated to large and complex MNEs – is a topic of growing interest for official statistic and is important to address the challenge of the impact of globalisation to the quality of economic statistics, ranging

from the business register all the way through to the quality and interpretation of the estimates of productivity. In terms of contributions, findings have firstly a practical implication for Istat, helping in defining the daily activities to be carried out by the Italian LCU team. Describing the Italian experience, the paper also aims to contribute to the international debate on the topic, providing suggestions for other NSOs wishing to set up similar-type units and affording similar challenges.

Even though this contribution consists of a synthesis of different studies made during the first phase of the project, some limits have been identified. Firstly, from an organisational perspective, part-time basis represents a good solution to create networks but for some members of the LCU Team, the working percentages on the project cannot be spread over the whole year and the ability to react quickly to the signals could be reduced due to the peaks and milestones of their other activities. Furthermore, restructuring cases are randomly distributed during the year and, accordingly, the LCU Team is not able to completely schedule all the activities to carry out well in advance. Even though limited by the internal organisation dispositions, obtaining an increase of the involvement percentages in the LCU for at least a part of the staff would maximise the results of the team.

Concerning future works, the improvement of inter-organisational relations represents the next main step to consider for the LCU Team. That implies the need to evaluate and setting up procedures to cooperate at different levels: from more informal ways within the same Directorate (in our case, Economic Statistics) to more structured protocols with other Directorates, such as Data Collection and National Accounts. For example, maximising synergies with the National Accounts team to deal with specific aspects such as economic ownership, contract manufacturing agreements, merchanting companies and enterprises with intellectual property product (IPP), would be crucial for a successful action. Furthermore, external relations with respondents and partnership with other national and international institutions need to be properly handled, in the first case to always obtain the necessary cooperation from the most relevant business groups and, in the second case, to have an increasingly complete view of the problematic aspects related to globalisation. Evaluating the degree of cooperation of the business groups in the target should represent the preliminary step to build solid relations with key respondents. Early studies show that top MNEs are keen to cooperate even though several difficulties emerged when asking questions related to specific globalisation issues. An effort in adopting a more user-friendly language, and less strictly statistical, should be made. Finally, it can be noted that even if limited in number, the EWS cases can determine an impact on the registration of variables and flows, highlighting the importance of improving relations with other countries.

## References

Alajääskö, P., M. Bujnowski, and B. Marola. 2018. "GDP/GNI: Challenges arising from globalisation - Eurostat's Multinationals (MNE) Pilot project". In *Proceedings of the 16<sup>th</sup> IAOS Conference*. OECD Headquarters, Paris, France, 19<sup>th</sup> - 21<sup>st</sup> September 2018.

Amante, S., S. Ambroselli, E. Bilotta, C. Orsini, and B. Gentili. 2018. "Large Cases Unit (LCU) monitoring activities: quick and complete assessment for complex restructuring cases". In *Proceedings of the Voorburg Group on Service Statistics*. Rome, Italy, 24<sup>th</sup> -18<sup>th</sup> September 2018.

Ambroselli, S. 2016. "A consistency unit for a consistent picture of the Italian economy". In *Proceedings of the 25<sup>th</sup> Meeting of the Wiesbaden Group on Business Registers. International Roundtable on Business Survey Frames.* Tokyo, Japan, 8<sup>th</sup> - 11<sup>th</sup> November 2016.

Bellini, G., N. Cecconi, L. De Gaetano, F. Monetti, P. Papa, and R. Ranaldi. 2018. "Centralizing data collection implementation: the Istat experience". In *Proceedings of the UNECE Conference of European Statisticians*. Workshop on Statistical Data Collection, Geneva, Switzerland, 10<sup>th</sup> - 12<sup>th</sup> October 2018.

Council Regulation (EEC) No 696/93, of 15<sup>th</sup> March 1993, on the statistical units for the observation and analysis of the production system in the Community.

Eurostat. 2017. *Early Warning System*. Luxembourg: Publications Office of the European Union.

Eurostat. 2010. "Business Registers. Recommendations Manual. 2010 Edition". *Eurostat Methodologies and Working Papers*. Luxembourg: Publications Office of the European Union.

Fazio, N., M. Murgia, and A. Nunnari. 2013. "The Business Statistical Portal: a new way of organising and managing data collection process for business surveys in Istat". In *Proceedings of the UNECE Seminar on Data Collection*. Geneva, Switzerland, 25<sup>th</sup> - 27<sup>th</sup> September 2013.

Hussain, M., R. Peltola, and S. Mahajan. 2019. "Measuring activities of multinational enterprise groups via large cases units". *EURONA: Eurostat Review on National Accounts and Macroeconomic Indicators*, N. 1/2019: 43-58.

International Financial Reporting Standard - IFRS. 2013. *IFRS 8 Operating Segments*. <u>http://www.ifrs.org</u>.

Italian National Institute of Statistics - Istat. 2016. *Istat's Modernisation Programme*. Roma: Istat. https://www.istat.it/it/files//2011/04/IstatsModernistionProgramme EN.pdf.

Statistics Canada, Enterprise Statistics Division. 2018. "Statistics Canada's Enterprise Portfolio Management (EPM) Program. Profiling and collecting data on large complex enterprises". In *Proceedings of the Voorburg Group*. Rome, Italy, 24<sup>th</sup> - 18<sup>th</sup> September 2018.

Stapel-Weber, S., and J. Verrinder. 2016. "Globalisation at work in statistics - Questions arising from the 'Irish case'". *EURONA: Eurostat Review on National Accounts and Macroeconomic Indicators*, N. 2/2016: 45-72.

Sturgeon, T.J. 2013. "Global Value Chains and Economic Globalization -Towards a new measurement framework". *Report to Eurostat*. Luxembourg: Eurostat.

United Nations, Statistics Division, Department of Economic and Social Affairs. 2019. Accounting for Global Value Chains: GVC Satellite Accounts and Integrated Business Statistics. New York, NY, U.S.: United Nations Publication.

United Nations, Statistics Division, Department of Economic and Social Affairs. 2013. "Guidelines on Integrated Economic Statistics". *Studies in Methods*, Series F, N. 108. New York, NY, U.S.: United Nations Publication.

United Nations Economic Commission for Europe - UNECE. 2015*a*. *Guide to Measuring Global Production*. Geneva, Switzerland: UNECE.

United Nations Economic Commission for Europe - UNECE. 2015b. Guidelines on Statistical Business Registers. Geneva, Switzerland: UNECE.

United Nations Economic Commission for Europe - UNECE, Organisation for Economic Co-operation and Development - OECD, and Eurostat. Group of Experts on the Impact of Globalization on National Accounts - GGNA. 2011. *The impact of globalization on national accounts*. New York, NY, U.S., and Geneva, Switzerland: United Nations.

van Delden, A., B. Lorenc, P. Struijs, and L.-C. Zhang. 2018. "Letter to the editor on statistical unit errors in business statistics". *Journal of Official Statistics*, Volume 34, Issue 2: 573-580.

Vennix, K. 2012. "The treatment of large enterprise groups within Statistics Netherlands". In *Proceedings of the Fourth International Conference on Establishment Surveys (ICES IV)*. Montréal, Québec, Canada, 11<sup>th</sup> - 14<sup>th</sup> June 2012.

# The task definition process for Istat Large Cases Unit

Simone Ambroselli, Sonia Amante, Elisabetta Bilotta, Barbara Gentili, Chiara Orsini, Fabiana Sartor <sup>1</sup>

## Abstract

The international debate on Large Cases Units (LCUs) has increased rapidly in recent years. Worldwide operations of Multinational Enterprises (MNEs) can deeply impact on the production of official statistics and setting up dedicated teams for handling MNEs has emerged as a key topic. However, the notion of such LCUs within a National Statistical Organisation (NSO) appears as a multifaceted concept. According to the experience of the Italian National Institute of Statistics - Istat, the main objective of this paper is to describe the steps followed in the first phase of the LCU project in defining the operational tasks needed to detect complex operations involving top MNEs that may affect both the quality of national statistics and the international comparability.

**Keywords:** MNEs, official statistics, LCU tasks, consistency, companies restructuring Early Warning System (EWS).

<sup>1</sup> Simone Ambroselli (ambrosel@istat.it); Sonia Amante (amante@istat.it); Elisabetta Bilotta (bilotta@istat.it); Barbara Gentili (bagentil@istat.it); Chiara Orsini (chorsini@istat.it); Fabiana Sartor (sartor@istat.it), Italian National Institute of Statistics - Istat.

The views and opinions expressed are those of the authors and do not necessarily reflect the official policy or position of the Italian National Institute of Statistics - Istat.

The authors would like to thank the anonymous reviewers for their comments and suggestions, which enhanced the quality of this article.

## 1. Introduction<sup>2</sup>

As worldwide operations of Multinational Enterprises (MNEs) can deeply impact on the production of official statistics, the setting up of dedicated teams for handling MNEs has recently emerged as a key topic. According to the Italian National Institute of Statistics - Istat Modernisation Programme (2016), a Large Cases Unit (LCU) team responsible for dealing with the inconsistencies of the top business groups operating in Italy has been established within the Directorate for economic statistics (DCSE)<sup>3</sup>. The main objective of the LCU is keeping the internal users – business statistics and national accounts experts – informed on the main restructuring events by identifying the effects of complex corporate restructuring at an early stage of the statistical production process.

The international debate on Large Cases Units (LCUs) has increased rapidly and in many different contexts. National experiences focussed on organisational aspects, such as the placement of the LCU within the National Statistical Organisation (NSO), but also analysed methodological topics, e.g. role, responsibilities, tasks and relationships (see among others, Statistics Canada, 2018; Wieser and Vennix, 2014; Connolly, 2011). Aiming at sharing common methods and best practices, some chapters on LCUs or similar activities have been included in international manuals, such as the United Nations Economic Commission for Europe (UNECE), Eurostat and the Organisation for Economic Co-operation and Development (OECD) guide on the Impact of globalisation on National Accounts (UNECE, 2011), the UNECE Guide to measuring global production in 2015 (UNECE, 2015) and the UNECE - Conference of European Statisticians (CES) Guide to sharing economic data (UNECE, 2020). The contribution of LCUs from different countries to GNI (Gross National Income) MNE group pilot exercise (2018-2019) could be regarded as a valuable experience, as this partnership set up within the European Statistical System was designed to verify the reliability of GNI considering globalisation issues.

<sup>2</sup> Although the article is the result of a joint work, the single paragraphs are authored as follows: Paragraphs 1, 2, 3, 4.2.1 and 5 by Simone Ambroselli; Paragraph 4.1.1 by Chiara Orsini; Paragraph 4.1.2 by Barbara Gentili; Paragraph 4.1.3 by Elisabetta Bilotta; Paragraph 4.2.2 by Fabiana Sartor; Paragraph 4.2.3 by Sonia Amante. Proofreading by Fabiana Sartor.

<sup>3</sup> Official name in Italian "Coordinamento delle unità economiche complesse" (Coordination of complex economic units).

In a broader perspective, the main purpose of an LCU team should be improving "the quality, consistency and coherency of data, although there can also be other beneficial impacts such as better use of resources and reducing the statistical reporting burden on MNE groups" (Hussain, *et al.*, 2019: 45). However, the notion of such LCU within a National Statistical Organisation (NSO) is a multifaceted concept especially regarding tasks to be performed. Even though generally, LCUs are not involved in all the actions (Hussain *et al.*, 2019), tasks may vary greatly and they may include management of specific surveys, data collection, contacts with respondents, monitoring activities and National Accounts analyses.

According to Istat experience, the main objective of this paper is to describe the steps followed in the first phase of the LCU project in defining the operational tasks needed to detect complex operations involving top MNEs that may affect both the quality of national statistics and the international comparability. Therefore, this work is presented in the context of official statistics, and specifically focusses on the LCU's (or similar units) organisation.

Complying with the LCU official mandate and acknowledging the findings of several analyses conducted in the starting phase of the project (Ambroselli *et al.*, 2020), the tasks have been defined according to the following two perspectives. Firstly, aiming at identifying the effects of complex corporate restructuring at an early stage of the statistical production process has been labelled as 'monitoring activities'; secondly, helping in dealing with the inconsistencies of the top MNEs in cooperation with the other statistical domains potentially affected, has been tagged as 'MNE analyses'.

Although the paper contributes to the international debate on LCUs, it mainly has practical implications for enhancing our understanding in defining the LCU operational tasks to carry out in our daily activity in order to achieve the set goals.

Nevertheless, the identification of the LCU role and responsibilities within the organisation must be considered as part of an adaptive process able to guarantee the best options in handling top business groups problems in both national and international context. In addition, the tasks should always be tuned considering other initiatives conducted within the NSO to avoid duplications at national level. The paper is organised as follows. Section Two intends to review different experiences concerning the responsibility of the LCU within the NSOs. The process followed by the Istat LCU team to define the tasks is discussed in Section Three while the description of the basic works that will be performed to fulfil the official mandate of the team are presented in Section Four. Discussion and conclusions are presented in Section Five.

#### 2. Responsibilities of the LCU

Even though several globalisation phenomena, such as merchanting, global manufacturing, processing abroad or international transactions in intellectual property products, have been analysed at international and European level and guidelines for recording them have been developed (Alajääskö et al., 2018), the need for boosting the efficiency in dealing with the measuring issues of global operations of MNEs is still relevant. Starting from 2016, Eurostat strategy, in addition to all the other projects and initiatives related to international measurement challenges, has been based upon two main actions. The first strategic action was developing a network, the so-called Early Warning System (EWS) (Limpach et al., 2018), in which Member States notify Eurostat of relocation or restructuring events concerning MNEs, while the second strategic action was promoting initiatives and adopt concrete measures to improve the consistent treatment for MNEs which are not relocating or restructuring. Concerning the former activity, it is possible to recognise what Stapel-Weber and Verrinder (2016) have called as an 'ex ante advice function' useful for the NSOs in the preparation of recording changes in MNEs structures. The latter action implies an integrated approach in which both business and macroeconomic statistics are involved: from the business registers to national accounts and Balance of Payment (BoP) passing through all the business statistics potentially involved. In particular, the strategy grounded on the use of the EuroGroup Register (EGR)<sup>4</sup> as starting point to have a common MNE structure helping in guaranteeing both internal and international coordination. Furthermore, all the initiatives already existing inside the NSOs, such as Profiling<sup>5</sup> and LCU work programmes, should be conceived to avoid duplication of work and progress faster<sup>6</sup>.

https://ec.europa.eu/eurostat/web/structural-business-statistics/structural-business-statistics/eurogroups-register

<sup>4</sup> EuroGroups Register (EGR) is the statistical business register of multinational enterprise groups having at least one legal unit in the territory of the EU or EFTA countries.

<sup>5</sup> Eurostat Business Registers Recommendation Manual (BRRM, 2010) as "a method to analyse the legal, operational and accounting structure of an enterprise group at national and world level in order to establish the statistical units within that group and their links, and the most efficient structures for the collection of statistical data".

<sup>6</sup> Guidelines from BSDG/DMES Joint Task Force on Globalisation concerning the MNE Pilot Exercise, June 2018. The Task Force was established in December 2016 by the Business Statistics Directors Group (BSDG) and Directors of Macro-Economic Statistics (DMES), in recognition of the need for a close cooperation between macroeconomic and business statisticians on globalisation-related issues.

After having defined the population of the groups in the target, according to Hussain *et al.* (2019) an LCU should focus on identifying and resolving the inconsistencies in primary statistical domains. In their contribution, they have identified a set of activities useful to treat inconsistencies at the early stage of the individual statistical processes before the dissemination. UNECE Guide to measuring global production (UNECE, 2015) offers another set of tasks derived from the results of a dedicated survey carried out by the Task Force on Global Production. In the first case, the authors have provided a contribution based on their personal assessment also considering that within the NSO usually there are other projects dealing with MNEs, such as profiling, while the second source have shown empirical results based on a review of national practices.

Combining the two sources, the following macro-groups could resume the potential LCU activities:

a) Contacts with the businesses.

Developing and maintaining regular communication and good working relationships with the selected MNE groups is an essential task for guaranteeing quality products. LCU could develop a new contact manager position, acting as a single contact point for the MNE group and other staff within the NSO. In that way, LCU experts could guarantee the necessary coordination between the NSO and respondents. On the respondent side, a contact manager could help in better understanding NSO statistical needs while, on the internal side, could enable the development of an integrated approach that offers mutual benefits and learning opportunities in dealing with business groups organisational changes. Holding meetings with companies could also be a specific task of the LCU, as a productive way to establish permanent relationships and to improve communication with the groups in the target. A face-to-face interaction could allow a smoother comprehension of the NSO requests and the adoption of a tailor-based approach.

b) Data collection.

In some countries, the LCU has been mainly related to data collection process. Different solutions have been adopted, such as choosing the most suitable method of data collection according to the group, eliminating duplicated questions and cutting the statistical reporting burden by reducing the number of questionnaires. LCU could also be directly involved in data collection in different ways, coordinating the activities for all the existing surveys, conducting specific surveys or developing new tools and methodologies starting from companies' accounts.

c) Monitoring.

Monitoring activities broadly concern the process to identify current and forthcoming issues in data provision and data analysis. This task could be done at different stages of the individual statistical process, from data collection to data analysis. According to what emerged from the international debate through evidence and recommendations, the cornerstone of a successful LCU programme is an efficient strategy for detecting problems before the dissemination and for handling them in an integrated perspective.

d) Data analysis

In several countries, LCU responsibilities involve consistency checks of different statistical and administrative data within and between the statistical domains. In some cases, data analysis also includes imputations and data adjustments. Generally speaking, the purpose should consist in analysing the complete set of data provided by the groups. More in detail, tasks could also include the provision of consistent data for business statistics as well as national accounts and balance of payments, and the exchange and reconciliation of mirror data with other countries.

e) Profiling

Delineation and classification of the statistical units of the main business groups operating in the country is a task that in some cases is associated to the LCU units. Profiling helps in achieving consistency over statistical domains through the complete definition of the business groups structures and the delineation of the enterprises.

f) Other tasks

Under the label of 'other tasks', each country includes very different tasks, such as the development of standards and tools. This activity is particularly useful to improve consistency rules and the knowledge of the team members concerning accounting and fiscal aspects, able to affect the organisation of the business groups and their capacity to provide meaningful statistical data. Based on these findings and according to its official mandate, Istat LCU has investigated on the main aspects to consider in defining the tasks of the team. Alongside external inputs, such as taking in charge EWS activities and following other international initiatives, business groups have been analysed considering their contribution to the National statistics and their degree of complexity (Ambroselli *et al.*, 2020). The team has adopted a new approach using an integrated overview of business groups, writing notes on restructuring events and case reports when needed. Furthermore, no additional data have been requested to the respondents in the classic form of a statistical survey. This procedure perfectly fits with the integrated system of Statistical Registers that has been recognised as one of the crucial elements of the Modernisation Programme.

### 3. Defining the tasks

Annually, top MNEs in the LCU target have been defined starting from the Italian Business Register<sup>7</sup> and the Italian Business Register of Enterprises Groups data<sup>8</sup>. The procedure starts at the business group level and two different orders for business groups are generated, by turnover and number of employees. After that, groups belonging to the top 200 ranking for both turnover and number of employees are selected. The third step consists in completing the list to obtain a total number of 130 business groups according to the turnover order only. Finally, other 10 worldwide top MNEs operating in the so-called digital economy, such as e-commerce and sharing economy, are added in the list even if their values are not remarkably high in Italy.

After defining the first list of the 140 groups in the target, the LCU team conducted preliminary studies according to its official purpose, which is keeping the internal users, such as business statistics and national accounts experts, informed on the main restructuring events by identifying the effects of complex corporate restructuring at an early stage of the statistical production process. The contribution of the enterprises in the scope was evaluated for the national Statistical Business Register (nSBR), monthly retail trade index and services turnover index. While aiming at investigating the complexity, the LCU team focussed on multi-business aspect and on the analysis of restructuring signals. Furthermore, interesting hints derived from some international initiatives coordinated by Eurostat or other international organisations to guarantee coordination among countries in dealing with inconsistencies of global MNEs.

Figure 3.1 synthesises the process followed for the definition of the tasks.

<sup>7</sup> https://www.istat.it/it/archivio/216767

<sup>8</sup> https://www.istat.it/it/archivio/234313



Figure 3.1 - Process for definition of the tasks

Source: Authors' elaboration

Input analyses produced several results that have been combined and evaluated. Tasks have been defined according to two main streams of activities: monitoring activities (Paragraph 4.1) and MNEs analyses (Paragraph 4.2). The former activities mainly aim at identifying the effects of MNEs complex restructuring at the early stages of the statistical production process, while the latter ones help in dealing with the inconsistencies of the top MNEs in cooperation with other statistical domains that are potentially affected.

Monitoring activities are carried out both internally, by sharing signals on major restructuring events and externally through the EWS network. Internal monitoring activities are planned on the information collected through the Istat Business Statistical Portal<sup>9</sup> and public news gathered from specialised newspapers and magazines. MNEs analysis generally followed signals activated by the monitoring activities. In other cases, cooperation with different Directorates may determine other specific investigations. Furthermore, signals coming from distinct statistical units belonging to the same business group could be an indication of a complex reorganisation that

<sup>9</sup> In Italian Portale statistico delle imprese https://imprese.istat.it/

need to be analysed more in detail. In such cases, an in-depth analysis is conducted and disseminated via specific reports. An assessment of the degree of cooperation in surveys through a group perspective may complete these case reports, which are normally updated when further information become available.

## 4. Day-to-day tasks

This section shows the tasks the LCU has implemented for its ordinary activity. According to the review of the activities generally allocated to the LCU (Paragraph 2), in our experience, day-to-day tasks are mainly referable to monitoring activities (c) with specific contributions for data analysis (d) both for national and international projects.

Working in cooperation with other Directorates, the LCU promoted preliminary projects concerning data collection (b) and contacts with the businesses (a). Speaking about data collection, in 2016 the entire process was entirely assigned to a new Directorate, thus the main contribution to this activity of the LCU is the evaluation of the degree of cooperation of the MNEs. As for the relations with the businesses, the LCU has established contacts also according to specific needs of the National Accounts Directorate. Furthermore, in terms of internal organisation, it is very important to underline that another Istat team is fully dedicated to profiling (e) and (EGR) activities.

## 4.1 Monitoring activities

International monitoring activities are carried out through the EWS network (Paragraph 4.1.1). Internal monitoring activities are based on the information collected through the Istat Statistical Business Portal, which is the web device focussed on retrieving data and managing contacts with all the enterprises (Paragraph 4.1.2), and on public news gathered from specialised newspapers and magazines (Paragraph 4.1.3).

## 4.1.1 The EWS flows

To understand the key concept of globalisation, various initiatives are bundled in EU to achieve coherence in methodologies and consistency in treating statistical issues. In line with European Statistical System principles, EWS aims to enhance statistical co-ordination, to promote integrated mechanism in exchanging information and to strengthen expertise in statistical areas. The interlinkages and integrated nature of the LCU are of crucial importance in ensuring that the purpose of the EWS is realised. Involving different expertise in the field of business statistics and National Accounts, the LCU team is able to provide the lens through which to manage MNE restructuring cases in an effective manner. As the LCU works towards wider range of MNE issues, on the one hand Istat correspondents for EWS are committed to triggering EWS Secretariat to anticipate the possible sizeable effects on national and European statistics due to the restructuring of an MNE. On the other hand, EWS Secretariat may trigger Istat correspondents for EWS, requiring them to provide certain information of MNE statistical profiles in order to develop a new approach in achieving statistical consistency while fostering synergies among Member States.

Since 2017 the LCU has been proactively participating in several EWS cases, by developing concise and effective reports concerning the relevant cases, as well as by ensuring responsive, inclusive and participatory decision-making processes in the field of statistics. Therefore, EWS activity is among the most advanced of those currently carried out by the LCU.

The triggering mechanism of the network produces inward and outward flows of information between EWS Secretariat and the LCU, which carries out monitoring activities of MNE statistics related issues while strengthening cooperation with EWS in a coherent and consistent manner. For both inward and outward flows, when MNE event is identified, which may include MNE mergers, acquisitions and restructuring, Istat correspondents for EWS, members of the LCU team, receive data and information from the experts working on the concerned statistical domains. Although generally limited to few variables or very specific situations, a mechanism of involvement of the potentially concerned domains is activated, similar to the way in which case reports are drafted (see Paragraph 4.2.1). In fact, in gathering relevant statistical information related to MNE profile, an internal consultation across various statistical domains is launched, by investigating Structural Business Statistics (SBS), Short-term Business Statistics (STS), Statistical Business Register (SBR), Foreign Affiliates Statistics (FATS), the Statistics of industrial goods and services (Prodcom) and Foreign Trade statistics, as well as National Accounts aggregates.

With regards to outward cases, once collected all relevant statistical data and indicators combined with public information about the Company such as Corporate website and newspapers, Istat correspondents trigger EWS Secretariat by informing that the restructuring case may affect other Member States and European statistics, accordingly. Istat correspondents provide the network with the MNE case overview containing the following information:

- detailed description of the restructuring case;
- statistical domains involved (*e.g.* STS, SBS, Business Registers, National Account aggregates);
- principal economic activity (NACE);
- estimated effects on the statistics;
- other countries which might be affected;
- timing of the restructuring;
- methodological assessment and proposal for statistical treatment;
- plans for communication to users.

The EWS secretariat establishes an *ad hoc* task force to collect information and to discuss the case involving the other Member States concerned. Integrating Member States opinions, EWS secretariat drafts Eurostat an interim report containing the information provided by the Member States and options to treat the MNE related statistics at the National and at the European levels. The mechanism of reaching consistency in statistical treatment of MNE case may require high-level institutional commitment, and effective and well-functioning inter-institutional coordination process, including, for instance, the involvement of the European Central Bank and National Central Banks. In close cooperation with the concerned correspondents, Eurostat produces a methodological note containing a summary of the case and an agreed methodological treatment to ensure a coordinated timing in the publication of first results and revisions. All the above-mentioned exchanging documents and information, including the methodological note and Member States opinions, are stored in a secure platform with an access only for correspondents differentiated by Case Group. Furthermore, as the agreed statistical treatment and consistent revisions may be applied to other cases, the repository of treated cases are available as anonymised explanatory notes containing the case description and the agreed methodological treatment for statistical unbalances available for all EWS members<sup>10</sup>.

<sup>10</sup> EWS Methodological summary notes https://ec.europa.eu/eurostat/web/economic-globalisation/early-warning-system

Even if EWS is based on the voluntary cooperation among National Authorities dealing with statistics, enhancing expertise on global statistics is crucial in developing effective procedures and in accelerating integrated approaches that leverage synergies across different dimensions of globalisation related issues. The network established by EWS provides a developing mechanism to support Member States to deal with statistical inconsistency due to globalisation events in a coherent manner.

#### 4.1.2 Signals from Istat Statistical Business Portal requests

Istat Statistical Business Portal (from hereon will be referred to as the Portal) is "a single entry-point for Web-based data collection from enterprises, which is at the same time an attempt to streamline the organisation and management of business surveys as a whole" (Fazio et al., 2013: 1). As a communication channel, the Portal guarantees the possibility to interact with respondents. In fact, it is "a tool for optimizing the bi-directional communication between Istat on one side and the companies involved in the statistical surveys on the other side" (Papa et al., 2018: 4). One of its key features consists in an integrated system that allows nSBR experts to manage the requests from enterprises in terms of changes (status and variables). The company can fulfil easily and quickly its obligations as well as monitoring the status of statistical requirements in real time; it can consult its situation in the nSBR and, if necessary, communicate changes receiving a response in real time. According to Consalvi et al. (2019), the tool has profoundly changed the methods of updating the nSBR from statistical sources and nowadays most of the updates of the register are collected through the Portal. Based on the management model adopted, an official application is activated for each request and a business register expert analyse and solve the case.

Concerning the LCU, monitoring activities aim at evaluating the signals from a business group point of view besides the enterprise level. In some cases, in fact, organisational changes in a business group consist in structural changes of several companies belonging to the same group.

From an operational point of view, all the requests sent via the Portal by the enterprises belonging to the business groups in the target are analysed. The full integration with the Business Register system allows the LCU team to automatically process data as identification codes, type of requests, variables affected and dates. Through this procedure, a provisional database containing signals on potential complex restructuring events involving the groups in the target is built up.

Starting from the target as defined in Section 3, for the period 2014-2019, 122 business groups (87 per cent of the reference population) and a total of 712 companies (22 per cent of the whole reference population in terms of legal units) communicate via the Portal showing a good propensity to use the tool. Enterprises belonging to the target groups are substantially stable (for example, about 84 per cent of the units are present in the years 2017 and 2018) but the incoming and outgoing movements in terms of single enterprises from a Business Register perspective can be considered as well significant.

YEAR	Number of requests
2014	58
2015	219
2016	273
2017	237
2018	187
2019	300
Total	1,274

Table 4.1 - LCU target enterprises, requests per year. Years 2014-2019

Source: Authors' elaboration based on Istat Statistical Business Portal requests

Table 4.2 - LCU target enterprises	, requests per variables.	Years 2014-2019
------------------------------------	---------------------------	-----------------

TYPE OF REQUESTS	Absolute value	Percentage value	
Mergers/split-offs	350	24.5	
Status	95	6.7	
Nace code (economic activity)	203	14.2	
Name	288	20.2	
Legal form	86	6.0	
Insolvency proceedings	43	3.0	
Address (administrative office)	363	25.4	
Total	1,428	100.0	

Source: Authors' elaboration based on Istat Statistical Business Portal requests

As shown in Table 4.1, 1,274 communications were received, 72 per cent of those cases were posted by the respondents. The rest of 28 per cent came from internal users, especially from short-term surveys. The requests were associated to 1,428 variables related to statistical classifications, demographic characteristics and structural changes (Table 4.2).

For LCU purposes, interesting signals concern structural changes, such as mergers, split-offs, requests for Nace code changes and communications about status and insolvency proceedings. In total, these requests amount to almost half (48 per cent) of the total. As previously mentioned, the main innovative aspect of our approach is related to the possibility to evaluate the requests according to a broader perspective (business group) in order to detect complex restructuring cases at an early stage of the statistical processes. For the most relevant cases, case reports are drafted and shared within business statistics and national accounts directorates (Paragraph 4.2.1).

#### 4.1.3 Signals from specialised newspapers and public sources

Evaluating the signals from specialised newspapers and magazines constitutes the second leg of the internal monitoring activities. News is analysed and organised by year in a dedicated repository. At the end of the year t, the records are further scrutinised, and three different groups are identified:

- a) events confirmed, codified and shared during the year t;
- b) events postponed to next year t+1;
- c) news not confirmed and not shared.

In terms of coverage, the activity concerns both the business groups in the LCU target and all the events that could involve Italian and foreign multinational companies in order to support the FATS statistics and the Statistical Business Register on Enterprise Groups processes.

Concerning the sources, the activity was mainly carried out by analysing the news reported by the specialised newspaper '*Il Sole 24 Ore*'. For the reference year 2019, the information sources treated were expanded: in addition to '*Il Sole 24 Ore*', the LCU used the 'Merger & Acquisition' report prepared by KPMG and the Newsletter distributed by the company Fineurop Soditic, specializing in mergers and acquisitions. This operation allowed an increase in the number and type of events reported.

The core objective is to identify the events that may involve the most significant enterprises or business Groups active in Italy and the impact for the statistics (acquisition projects, mergers, opening of new sites, liquidation, expansion projects abroad).

Before sharing the files, the confirmed events are checked and coded. This last phase must be as accurate as possible since the names of the companies involved must be traced back to the identification codes otherwise the information could not be used.

The undefined events are divided between those that have not materialised and those still being defined or completed. The events postponed are recorded separately so that they can follow their evolution over the following year.

For both groups of events confirmed and postponed, two distinct dissemination files are shared within Istat. Excel format to share the cases with a reduced number of variables has been chosen in order to obtain files that are easily and quickly interpretable. Content of both files is described in Table 4.3.

The report for the year 2019 created through analysing data on daily basis, summarises information on cases concerning companies and MNEs mainly involved in mergers/split-offs and sales or acquisitions of subsidiaries. While considering that the number of cases analysed is strictly dependent on the underlying economic reality and highly variable from year to year, for 2019 the number of events analysed compared to previous years has been more than double, also because new information sources have been fully exploited.

Starting from a total of 449 events analysed, those defined during 2019 have been 376 (Table 4.4) while in 3 cases their conclusion took place in 2020. In 37 cases the events have been postponed to the following year, while 7 cases needed further investigations. The latter situations have concerned very complex restructuring operations of large Italian groups. Finally, we found that in 26 cases the reports concerned only announced events that did not materialise.

Results in terms of type of monitored events for the year 2019 are shown in Table 4.5. Acquisitions of subsidiaries (or entire business groups) account for three quarters of the total, while the rest concern MNEs reorganisations both

internal and related to other parties (mergers, split-offs, sale and cessation of activities).

Dissemination file 1 - confirmed events	Dissemination file 2 - postponed events
Progressive identification code	Progressive identification code
Identification code main statistical unit	Identification code main statistical unit
Name statistical unit	Name statistical unit
Type of event	Type of event
Extended description of the event	Extended description of the event
Other Units' codes	
Other Statistical units' names	
UCI* name or GGH** name	
UCI or GGH country	
Group name involved in the event	
Survey on foreign-controlled companies resident in Italy (Inward FATS)/or foreign affiliates abroad controlled by Italy (outward FATS)	Survey on foreign-controlled companies resident in Italy (Inward FATS)/or foreign affiliates abroad controlled by Italy (outward FATS)
Indication of the different domains interested in the survey	
Territorial impact (IT/FRGN)	
Date (certain) of the event or reference year	
	Expected date of the event
Other notes	Other notes
Source: Authors' elaboration	

Table 4.3 - Content of the dissemination files

\*UCI: Ultimate Controlling Institutional Unit.

\*\*GGH: Global Group Head.

#### Table 4.4 - Monitoring results. Year 2019

EVENTS	Number
Events carried out during the year	376
Events of large groups not yet defined during the year	7
Events postponed to the following year	37
Events closed in the following year	3
Events not occurred	26
Total events analysed	449

Source: Authors' elaboration

TYPE	Percentage value
Acquisitions	75.3
MNE internal events	10.7
Sale / cessation of activitieS	7.4
Mergers/split-offs	6.6
Total events carried out during the year	100.0

Table 4.5 -	Type of the	events	carried	out	durina	the	vear.	Year	2019
10010 4.0 -	Type of the	C V CIIILO	curricu	out	aaring	uic	your.	i cui	2010

Source: Authors' elaboration

The analysis of 2019 data reveals that approximately 24 per cent of the events concerned acquisitions of Italian companies by foreign multinationals. In Italy took place 43 per cent of the operations (acquisitions, mergers, opening new offices), while 19 per cent of the operations were managed abroad by Italian multinationals; finally, approximately 14 per cent of the operations were carried out between foreign groups.

## 4.2 MNEs analyses

Case reports (Paragraph 4.2.1) and the evaluation of the degree of cooperation to the surveys from a business group perspective (Paragraph 4.2.2) represent the main outputs labelled as 'MNEs analyses'. Other *ad hoc* analyses are presented at the end of the Paragraph 4.2.3.

### 4.2.1 Case reports

Case reports represent the main tool the LCU team uses for sharing information concerning groups in the target, such as restructuring, changes of ownerships and impact on statistical units. The purpose of the task is carrying out concrete actions for the development of a coherent set of information useful to improve a consistent treatment of complex MNEs within different business statistics. Without prejudice to the autonomy of each production process, the aim of LCU reports is to facilitate a shared solution and the application over time of the same solutions for similar cases. Cooperation in MNE data treatment is a key task for the success of the LCU.

A case report is provided when MNEs reorganisations seem to be able to create 'breaks' in the statistical outputs. The urgency of the case is closely linked to the needs of the involved domains. Currently, the variables under control are related to classification (Nace code), turnover and employment, as well as a general assessment of the structure of the groups in terms of statistical units. Checks are carried out by considering statistical data and Financial Statements (Annual Reports and Consolidated Accounts). By the report, an internal consultation of the concerned statistical domains is opened. Firstly, the assessment is made considering the effects produced (or that will be produced) in the short term and therefore able to have an impact on STS and SBR domains. At the same time, the other primary statistics (SBS, FATS, PRODCOM, Foreign Trade statistics) are involved to facilitate the most homogeneous treatment of the events in a wider perspective. For the same reason, a dedicated National Accounts team for the treatment of the globalisation process is also involved for assessing the need to investigate specific aspects not captured by primary statistics but necessary for the compilation of national accounts.

Operationally, the reports derive from signals collected via monitoring activities (Paragraph 4.1) or follow some specific needs emerged inside the NSO. Reporting is an iterative process that engages people in charge for the different statistical domains potentially involved. Comments and suggestions are collected via a dedicated email box or, preferably, writing using a track changes function inside a common protected area.

#### Table 4.6 - Case reports: checklists

		Checklist 1		
Statistical Units	Impact		To do	
Groups	Yes	No	Yes - go to Checklist 2	
Enterprises	Yes	No	Yes - go to Checklist 3	
Local Units	Yes	No	Yes - go to Checklist 4	
Checklis	st 2 – Groups		_	
Variables	Imp	oact		
Existing control links	Yes	No	_	
New control links	Yes	No		
UCI (GGH)	Yes	No		
Complex reorganisation	Yes	No	_	
	Chec	klist 3 – Enter	prises	
Variables			Imp	act
Variables Enterprise events (mergers; tak	eovers/break-ups; sp	lit offs)	Imp Yes	act No
Variables Enterprise events (mergers; tak Nace	eovers/break-ups; sp	lit offs)	Imp Yes Yes	act No No
Variables – Enterprise events (mergers; tak Nace Employment	eovers/break-ups; sp	blit offs)	Imp Yes Yes Yes	act No No No
Variables Enterprise events (mergers; tak Nace Employment Status	eovers/break-ups; sp	olit offs)	Imp Yes Yes Yes Yes	act No No No No
Variables Enterprise events (mergers; tak Nace Employment Status Headquarters	eovers/break-ups; sp.	olit offs)	Imp Yes Yes Yes Yes Yes	act No No No No No

Variables	Impact		
Headquarters	Yes	No	
Local units (mainly factories)	Yes	No	
Others	Yes	No	

Source: Authors' elaboration

Reports contains the following parts:

- description of the event and data available;
- type of signal: internal/external;
- source of signal: monitoring activities; specific needs inside Business Statistics; input from other Directorates; international projects;

- evaluation of the impact for the statistical production in terms of statistical units (business groups, enterprises, legal units and, partially, local units) and variables (mainly, employment, turnover, economic activity classification and location);
- outcome of the preliminary analysis: the potential impact and the proposals/decisions for the involved domains;
- list of the involved statistical units (names and Register codes);
- contributors and contact persons for the case;
- version.

Even though the task is composed of a continuum of operations and some phases can be contextual or iterative, the following macro-steps have been defined to follow in drafting the reports, not necessarily in a rigid sequence. Classifying the signals and collecting further information and data concerning the case constitute the first macro-step of the activity and allow a quick evaluation of the complexity of the case. The actual handling of the case starts immediately after the first assessment and represents the core stage of the entire process: the urgency of the case is assessed, the inconsistencies are identified and described and the impact on the main users is predicted at a sufficiently detailed level of information. Basically, the evaluation should enable to answer the checklists of statistical units and variables showed in Table 4.6.

In this phase, the need for contacting the business could emerge and members of the LCU provide the necessary steps to organise the contact properly. At the end of this phase, information on the expected effects and operational proposals and/or decisions are added in the report.

The subsequent step consists in sharing the first draft of the report in order to obtain the involvement in the analysis of the thematic experts. In that way decisions to be taken could be refined and approved.

In the final step of the process, the final version (rev. 1) of the report is shared with all the statistical domains even if not immediately affected. It is important to consider that in our perspective the responsiveness in analysing the signals concerning complex cases is of primary interest even if new versions of the report could be made in case of relevant updates.

#### 4.2.2 Business groups: degree of cooperation and statistical burden

A high response rate guaranteed by the top MNEs is a key aspect to obtain accurate statistical outputs. Either as a complement of case reports or as an independent analysis, one of the tasks of the LCU team concerns studying how business groups in the target cooperate in surveys, supplying tables for resuming and comparing data. For this purpose, a group perspective has been adopted for all analysis on cooperation.

While providing significant timely data and high-quality statistics, NSOs should always consider the need to minimise the reporting burden. Principle 9 of the "European Statistics Code of Practice For the National Statistical Authorities and Eurostat (EU statistical authority)" by Eurostat (2011 and revised edition 2017) states as "the reporting burden should be proportionate to the needs of the users and should not be excessive for respondents. The statistical authority monitors the response burden and sets targets for its reduction over time". Business groups generally express concerns over the statistical burden they have to bear while official statistical producers have to face increasing demands for information and a remarkable effort to be compliant with EU Regulations. Nowadays blending these opposing needs offers a compelling challenge to NSOs.

According to Istat internal organisational, each survey manager and the Data Collection Directorate are in charge of the burden evaluation, especially regarding the contents of the questionnaires. Furthermore, for each survey a dedicated team of experts check if the submitted questionnaires are filled in properly. Collaboration between the Data Collection Directorate and the LCU team allows measuring the actual cooperation of large business groups in official statistics, serving two main purposes. First, this analysis helps to assess the availability and the quality of the data (collected or estimated) used for consistency checks; secondly, it allows to evaluate the feasibility of requesting further cooperation (direct contacts,) especially for globalisation related issues. Furthermore, updates and new versions of the reports (see Paragraph 4.2.3) allow coverage assessment of data collection after restructuring.
ECONOMIC BUSINESS SUR- VEYS	Business group enterprises								
	Ent.1	Ent.2	Ent.3	Ent.4	Ent.5	Ent.6	Ent.7	Ent.8	Tot
OCC1	Х		Х			Х	Х		4
RS1	х	Х	Х	Х	Х	Х			6
PRODCOM	Х		Х	Х		Х	Х		5
PPI_D	Х					Х			2
CIS	х		Х		Х	Х	Х		5
FID_MAN	Х		Х						2
ICT	х		Х	Х	Х	Х	Х		6
RCL-SES	Х		Х			Х	Х		4
SCI	Х		Х	Х		Х	Х		5
FATT	Х		Х				Х		3
PMI	_			Х					1
IPI	Х		Х	Х			Х		4
IULGI	Х		Х	Х		Х	Х		5
INWARD	Х	Х	Х	Х	Х	Х	Х		7
IS	Х		Х	Х	Х	Х	Х		6
PREIMP				Х					1
CENS	Х		Х	Х		Х	Х		5
Tot	15	2	14	11	5	12	12	0	71

## Table 4.7 - Istat business surveys involving enterprises of the examined MNE group, 2017-2019

Source: Authors' elaboration on Istat data

STS in grey

Labour Force Survey (OCC1); Research & Development Survey (RS1); Community Production Survey (PRODCOM); Producer Price Survey – domestic market (PPI\_D); Community Innovation Survey (CIS); Business Confidence Survey of Manufacturing Industry (FID\_MAN); Information, Communications and Technology Survey (ICT); Structure of Earning and Cost of Labour Survey (RCL-SES); SBS Total Survey on large enterprises (SCI); Industrial New Orders and Turnover Survey (FATT); SBS Sample Survey on small and medium enterprises (PMI); Industrial Production Survey (IPI); Survey on Local Units of Large Enterprises (IULGI); Activities of Foreign Enterprises in Italy (INWARD); Survey on International Sourcing (IS); Import Prices Survey (PREIMP); Businesses and Services Census (CENS).

When monitoring signals and case reports reveal that a group restructuring involves more enterprises and direct contacts, in order to get a clearer picture, the LCU team produces specific reports as showed in the following business case.

The business group considered for this example is an important foreignowned MNE operating in Italy. Although its core business unit focusses on a productive activity, other enterprises of the group operate in several different fields orbiting the main mission. Using the Portal, we could track down the list of surveys in which enterprises belonging to the MNE group were involved. Analysing the response rate for each enterprise of the group to different surveys has helped evaluate the weight of the actual burden on the whole group and to establish the degree of cooperation of the MNE.

The study revealed that the considered business group has 8 enterprises engaged in the Istat surveys (both structural and short-term) in 2017-2019. Table 4.7 shows the results of the investigation. The larger enterprises (1 and 3) were involved in almost all the surveys, while RS1 Survey, ICT Survey, INWARD Survey and IS Survey engaged more than 5 (out of 8) enterprises of the group.

The response rate to the Istat questionnaires has been rather high for six enterprises while in one case only half of the surveys has been filled. However, after the contact, percentages have increased for all the enterprises belonging to the group.

Direct contacts with the business have allowed also discovering that for the same period, the enterprises belonging to the group have received a relevant number of other surveys from National Authorities or other Public Administrations (9 mandatory out of 16).

A first attempt to extend this analysis to other MNEs shows that the majority of the MNEs tend to cooperate actively in surveys, therefore their average response rate is fairly satisfactory. However, MNEs often report having troubles disaggregating data into small geographic areas and by enterprise. The analysis of the statistical burden on MNEs revealed that notwithstanding the effort required to satisfy the demand for information, MNEs tend to cooperate in data collection. Overall, fulfilling these obligations has a cost for both businesses and NSOs, as excessive burden affects the quality of gathered data (redundant information and overlap between business surveys) and the efficiency of data collection process.

## 4.2.3 Updates and other contributions

Last stream of research concerning 'MNEs analyses' consists of cases updates and other specific contributions made in cooperation with other Directorates mainly for international projects or to establish direct contacts with the businesses.

As seen in Paragraph 4.2.1, when necessary, a case is updated according to new information, data or experts' evaluations (business statistics and national accounts). In these cases, the LCU team generally opts for drafting a new version of the report.

In other cases, as seen in Paragraph 4.1.3, some corporate reorganisations can be very complex and require very long time. The time that passes from the announcement of an event gathered from newspapers to its actual and complete realisation may vary: sometimes events close very soon, while every now and then events can last for months, especially when, for example, the authorisations of the National Competition Authorities involved are necessary. In these cases, the LCU provides some specific contributions that briefly summarised the main aspects of the ongoing operations, the current situation and a time estimation concerning the conclusion of the restructuring. Furthermore, internal identification codes are retrieved in order to connect information to the statistical units. A newsletter is prepared and disseminated to business statistics and national accounts domains. In fact, in this preliminary phase it is important for expert to think over the potential impact that the event, once concluded, could have on the production process which is affected. The newsletter is shared at non-regular intervals when relevant news is available. Given the public nature of the information collected and disseminated, there are no confidentiality issues and email can be used for dissemination.

In our experience, cases disseminated by this tool have covered a wide variety of aspects concerning the main business groups acting in Italy, such as governance, expansion abroad of Italian business groups, organisational changes in terms of legal units and company crises.

Other non-regular contributions have been made in cooperation with National Accounts Directorate in order to help define of specific issues of macroeconomic statistics. Usually, the LCU contribution is made at a preliminary stage, complementing the information from the profiling activity with a general assessment of the business group regarding ongoing events that could substantially impact on primary statistics outputs and, as a consequence, on national accounts aggregates.

Finally, although to a lesser extent, other preliminary analyses have been carried out in order to get a complete overview of the activities of the business group before contacting the businesses. The LCU members contribute to summarise the context, evaluating the cooperation of the single entities belonging to the business group (as seen in the previous section) and looking for the best strategies to establish or maintain the direct cooperation with the respondents.

## 5. The LCU role: an adaptive process

In order to define the operational tasks needed to detect complex operations concerning the main business groups acting in Italy and able to affect both the quality of national statistics and the international comparability, the LCU team conducted several studies in the first phase of the project.

Tasks have been defined according to two different streams. The first set of activities consists in monitoring the MNEs to identify current and forthcoming problems in data provision and data analysis as a consequence of restructuring events. The second set of activities analyses top MNEs to manage data inconsistencies, alerting other potentially affected statistical domains. Even though a logical sequence can be recognised considering monitoring activities as upstream in forming the informative base from which in-depth and other contributions can be extrapolated, relationships could also follow other ways. Therefore, most of the tasks could run parallel with continuous interrelations in order to guarantee the necessary flexibility to the LCU project.

Three different small teams within the LCU follow the three different streams of research belonging to monitoring activities macro-group. For internal monitoring activities (Istat Statistical Business Portal; specialised newspapers and magazines) two different repositories are maintained and mutual exchange of data exist allows comparing the results by internal identification codes and events descriptions and typologies. Both the internal monitoring tasks can feed the international monitoring activities that are carried out through the EWS network. In fact, triggering complex restructuring cases that involve more EU Countries is a task of the LCU team both for inward and outward flows of information between EWS Secretariat and Istat. In the former case, EWS correspondents, members of the LCU team, receive the cases from the EWS Secretariat and then activate an internal mechanism in order to obtain the necessary feedback to define the Italian position concerning the case. In some of these cases, a re-evaluation of the signals gathered from the Portal could be necessary. For outward flows, the EWS correspondents, being aware of a restructuring event able to affect other EU Countries mainly by exploiting the results of internal monitoring activities, trigger the EWS secretariat to activate an EWS case discussion.

Relations between monitoring and MNEs analyses are mainly one-way even if some contributions on MNEs may constitute the starting point for an EWS outward case. Despite the high number of signals collected through the upstream activity, only few cases have the necessary characteristics - in terms of impact or complexity of the restructuring event - for drafting and disseminating a case report or other kind of contributions. As seen in the previous sections, case reports guarantee the development of a coherent set of information useful to improve a consistent treatment of complex MNEs within different business statistics, however the process is time-consuming and requires the involvement of several statistical domains. For these reasons, only very relevant cases have been chosen for reporting and follow-ups. Cases updates and other contributions, as well as the production of outputs concerning the actual cooperation to the survey according to a business group perspective, are more frequent but strictly dependent on external factors, such as the dynamics of the economic context and the needs of other projects. Operationally, the three tasks within the MNEs analyses macro-group are related by two-ways relationships.

Nevertheless, according to our experience, the identification of the LCU role and responsibilities within the organisation must be considered as part of an adaptive process mainly for the following reasons.

Firstly, the tasks should be always tuned considering the other initiatives conducted within the NSO to avoid duplications at national level. In general, crosscutting activities are difficult to accept within an organisation and the LCU is not an exception. The presence of several experts with different skills and engaged in other processes, has proved to be a solution suited to the context but, at the same time, has limitations. In such a situation, strengthening the endorsement by the top managers of the NSO may become essential to survive and guarantee the best options in handling top business groups problems for both a national and an international perspective. However, the ever-increasing interest on the topic of Eurostat and other international organisations has helped us in explaining and adjusting the tasks according to a wider perspective.

Secondly, the adaptation should also consider the labour force allocation balancing different needs, when possible. Currently, resources have been considered as sufficient for the LCU target definition, monitoring activities and for participating in the EWS network. However, with regard to the MNEs analyses, more resources are needed. In fact, concerning the case reports, very good relationships have been established with the thematic experts obtaining a high degree of involvement, but LCU resources are not sufficient to ensure greater production and to follow the consistent implementation of the proposed solution for the cases.

In order to respond adequately to internal and external impulses and to be ready to partly modify its tasks or taking other responsibilities in charge, LCU future plans should also consider training and development of synergies.

Concerning the former, the training of the LCU staff should also be considered as an ongoing process. Promoting the enlargement of the competences for each member of the team could help in enhancing the understanding of the company side not only in terms of structure of legal units but also considering their actual internal organisations for aspects such as degree of internationalisation, regulatory changes that determined internal reorganisations and international accounting standard rules. Increasing the knowledge about MNEs operations in terms of legal aspects, such as Regulations and antitrust rules, International Accounting Standards and group taxation rules, emerged as a key urgent aspect to be developed by the LCU team. Furthermore, as international aspects tend to be very demanding, the LCU team might need to improve its knowledge on international networks operations and its skills to better deal with issues and obstacles (legal, administrative, statistical, technical and cultural) associated with international data sharing.

Concerning the latter, in order to exploit the organisational synergies, the LCU team has planned to collect and effectively keep track of Data Collection and National Accounts Directorates' needs, primarily for improving the effectiveness of direct contacts with the business groups and investigate on globalisation effects caused by activities of MNEs and their coherent and systemic reflection in statistical data. Cooperation with other national authorities, such as the National Central Bank, should also be favoured in order to both enlarge the sources available and include other statistical domains in the analysis. Finally, networking and sharing experiences on LCU work should consider both EU and non-EU countries. In the latter case, international organisations could promote actions to share best practises but also bilateral

studies to examine specific situations. Such practical applications would be more supportive to complete the analyses considering that MNE activities are not limited to the EU.

In conclusion, the article describes the practical implementation in Istat of the LCU tasks in the first years of the project. According to organisational aspects and preliminary studies, during this first phase, the main task of the Italian LCU has been providing information on important restructuring events concerning MNEs to internal users. In order to make the most out LCU work, future activities should mainly boost the identification process to define the restructuring effects on national statistics, assessing the consistency of MNE data from primary statistics and either cooperating with other statistical domains in dealing with globalisation issues or providing backing up for experts' individual decisions. Eventually, arrangements should be made to implement and subsequently improve the use of automatic checking procedures to detect and manage complex issues.

## References

Alajääskö, P., M. Bujnowski, and B. Marola. 2018. "GDP/GNI: Challenges arising from globalisation - Eurostat's Multinationals (MNE) Pilot project". In *Proceedings of the 16<sup>th</sup> IAOS Conference*. OECD Headquarters, Paris, France, 19<sup>th</sup> - 21<sup>st</sup> September 2018.

Amante, S., S. Ambroselli, E. Bilotta, C. Orsini, and B. Gentili. 2018. "Large Cases Unit (LCU) monitoring activities: quick and complete assessment for complex restructuring cases". In *Proceedings of the Voorburg Group on Service Statistics*. Rome, Italy, 24<sup>th</sup> - 18<sup>th</sup> September 2018.

Ambroselli, S. 2016. "A consistency unit for a consistent picture of the Italian economy". In *Proceedings of the 25<sup>th</sup> Meeting of the Wiesbaden Group on Business Registers. International Roundtable on Business Survey Frames.* Tokyo, Japan, 8<sup>th</sup> - 11<sup>th</sup> November 2016.

Ambroselli, S., S. Amante, E. Bilotta, B. Gentili, B. Iaconelli, L. Mattei, C. Orsini, F. Sartor, and I. Straccamore. 2020. "Handling Multi-National Enterprises: Istat's early experience in Large Cases Unit". *Rivista di statistica ufficiale* N. 1/2019. Roma: Istat.

Bellini, G., N. Cecconi, L. De Gaetano, F. Monetti, P. Papa, and R. Ranaldi. 2018. "Centralizing data collection implementation: the Istat experience". In *Proceedings of the UNECE Conference of European Statisticians*. Workshop on Statistical Data Collection, Geneva, Switzerland, 10<sup>th</sup> - 12<sup>th</sup> October 2018.

Connolly, M. 2011. "CSO's Large Cases Unit – A strategy for dealing with multinationals and globalisation". In *International Statistical Institute - ISI, Proceedings of 58<sup>th</sup> World Statistical Congress*. Dublin, Ireland, 21<sup>st</sup> - 26<sup>th</sup> August 2011.

Consalvi, M., B. Gentili, and A. Speranza. 2019. "The new role of the SBR within the Italian Business Portal". In *Proceedings of the Meeting of the Group of Experts on Business Registers*. UNECE, Eurostat and OECD. Geneva, Switzerland, 30<sup>th</sup> September - 2<sup>nd</sup> October 2019 Geneva. Switzerland.

Council Regulation (EEC) No 696/93, of 15<sup>th</sup> March 1993, on the statistical units for the observation and analysis of the production system in the Community.

Eurostat, and European Statistical System - ESS. 2017. European Statistics Code of Practice for the National Statistical Authorities and Eurostat (EU statistical authority). Luxembourg: Publications Office of the European Union.

Eurostat. 2017. *Early Warning System*. Luxembourg: Publications Office of the European Union.

Eurostat. 2010. "Business Registers. Recommendations Manual. 2010 Edition". *Eurostat Methodologies and Working Papers*. Luxembourg: Publications Office of the European Union.

Fazio, N., M. Murgia, and A. Nunnari. 2013. "The Business Statistical Portal: a new way of organising and managing data collection process for business surveys in Istat". In *Proceedings of the UNECE Seminar on Data Collection*. Geneva, Switzerland, 25<sup>th</sup> - 27<sup>th</sup> September 2013.

Hussain, M., R. Peltola, and S. Mahajan. 2019. "Measuring activities of multinational enterprise groups via large cases units". *EURONA: Eurostat Review on National Accounts and Macroeconomic Indicators*, N. 1/2019: 43-58.

Italian National Institute of Statistics - Istat. 2016. *Istat's Modernisation Programme*. Roma: Istat.

https://www.istat.it/it/files//2011/04/IstatsModernistionProgramme\_EN.pdf.

Limpach, S., A. Götzfried, H. Linden, and J. Verrinder. 2019. "The Early-Warning System". In *International Statistical Institute - ISI, Proceedings of the 62<sup>nd</sup> World Statistics Congress*. Kuala Lumpur, Malaysia, 18<sup>th</sup> - 23<sup>rd</sup> August 2019.

Statistics Canada, Enterprise Statistics Division. 2018. "Statistics Canada's Enterprise Portfolio Management (EPM) Program. Profiling and collecting data on large complex enterprises". In *Proceedings of the Voorburg Group*. Rome, Italy, 24<sup>th</sup> - 18<sup>th</sup> September 2018.

Stapel-Weber, S., and J. Verrinder. 2016. "Globalisation at work in statistics - Questions arising from the 'Irish case'". *EURONA: Eurostat Review on National Accounts and Macroeconomic Indicators*, N. 2/2016: 45-72.

United Nations Economic Commission for Europe - UNECE. 2020. *Guide to Sharing Economic Data in Official Statistics*. Task Force on exchange and sharing of economic data. Geneva, Switzerland: United Nations.

United Nations Economic Commission for Europe - UNECE. 2015. *Guide to Measuring Global Production*. Geneva, Switzerland: UNECE.

United Nations Economic Commission for Europe - UNECE, Organisation for Economic Co-operation and Development - OECD, and Eurostat. Group of Experts on the Impact of Globalization on National Accounts - GGNA. 2011. *The impact of globalization on national accounts*. New York, NY, U.S., and Geneva, Switzerland: United Nations.

Wieser, M., and K. Vennix. 2014. "Consistency and profiling of large enterprise groups within Statistics Netherlands". In *Proceedings of the 24<sup>th</sup> Meeting of the Wiesbaden Group on Business Registers - International Roundtable on Business Survey Frames*. Vienna, Austria, 15<sup>th</sup> - 18<sup>th</sup> September 2014. The *Rivista di statistica ufficiale* publishes peer-reviewed articles dealing with crosscutting topics: the measurement and understanding of social, demographic, economic, territorial and environmental subjects; the development of information systems and indicators for decision support; the methodological, technological and institutional issues related to the production process of statistical information, relevant to achieve official statistics purposes.

The *Rivista di statistica ufficiale* aims at promoting synergies and exchanges between and among researchers, stakeholders, policy-makers and other users who refer to official and public statistics at different levels, in order to improve data quality and enhance trust.

The *Rivista di statistica ufficiale* was born in 1992 as a series of monographs titled "*Quaderni di Ricerca Istat*". In 1999 the series was entrusted to an external publisher, changed its name in "*Quaderni di Ricerca - Rivista di Statistica Ufficiale*" and started being published on a four-monthly basis. The current name was assumed from the Issue N. 1/2006, when the Italian National Institute of Statistics – Istat returned to be its publisher.

La Rivista di statistica ufficiale pubblica articoli, valutati da esperti, che trattano argomenti trasversali: la misurazione e la comprensione di temi sociali, demografici, economici, territoriali e ambientali; lo sviluppo di sistemi informativi e di indicatori per il supporto alle decisioni; le questioni metodologiche, tecnologiche e istituzionali relative al processo di produzione dell'informazione statistica, rilevanti per raggiungere gli obiettivi della statistica ufficiale.

La Rivista di statistica ufficiale promuove sinergie e scambi tra ricercatori, stakeholder, policy-maker e altri utenti che fanno riferimento alla statistica ufficiale e pubblica a diversi livelli, al fine di migliorare la qualità dei dati e aumentare la fiducia.

La Rivista di statistica ufficiale nasce nel 1992 come serie di monografie dal titolo "Quaderni di Ricerca Istat". Nel 1999 la collana viene affidata a un editore esterno, cambia nome in "Quaderni di Ricerca - Rivista di Statistica Ufficiale" e diventa quadrimestrale. Il nome attuale è stato scelto a partire dal numero 1/2006, quando l'Istituto Nazionale di Statistica - Istat è tornato a esserne l'editore.